# APPENDIX A



## Administrative Conference of the United States

### **REGULATIONS.GOV AND THE FEDERAL DOCKET MANAGEMENT SYSTEM**

December 1, 2018

### Todd Rubin Administrative Conference of the United States

This report was prepared for the consideration of the Administrative Conference of the United States. The opinions, views and recommendations expressed are those of the author and do not necessarily reflect those of the members of the Conference or its committees, except where formal recommendations of the Conference are cited.

Contents
----------

Introduction	2
Part I. What FDMS/Regulations.gov Looks Like and Its Core Functionalities	3
Part II: How FDMS/Regulations.gov is Governed and Funded	4
Part III: How FDMS/Regulations.gov Came Into Being	5
Part IV: Participation in FDMS/Regulations.gov Facilitates Compliance with the E- Government Act of 2002	6
Part V: The Key Challenges Underlying Regulations.gov	7
A. Users Find It is Difficult to Consistently and Reliably Search for Rulemaking Materials	8
B. E-Dockets are Not Reliably Informative	13
Part VI: Solutions	. 14
Acknowledgments	. 18

#### Introduction

Federal regulations (hereinafter "rules") affect nearly all aspects of our lives. The Administrative Procedure Act, the statute that governs a large part of the federal rulemaking process, generally requires agencies to give the public notice of, and the opportunity to comment on, rules they are considering issuing.<sup>1</sup> The main purposes of this mandated opportunity for public input are to facilitate agencies' access to widely dispersed information and, if necessary, to help regulators rethink critical assumptions about rulemaking proposals, all with the goal of improving the quality of rulemaking.

To submit an informed comment, potential commenters need to be able to at least: 1) access the proposed rule and the agency's justification for it; 2) access materials upon which the agency substantially relied to develop the proposed rule; and 3) understand the rationale by which the agency made its decision.<sup>2</sup> Commenters should also be able to access other comments that may have been submitted on the proposed rule in time to submit responsive comments, to the extent this is possible. Members of the public, especially those who are subject to the rule, should be able easily to determine whether further action has been taken on the proposed rule and, when a final rule has been issued, to access the rule and all materials, including public comments, that informed its development.

Historically, it has been a challenge for many people to understand the rulemaking process and to access these rulemaking materials. Before the internet was widely available, members of the public interested in reading materials in a rulemaking docket (e.g., supporting materials and other comments submitted) needed to go to the agency and schedule an appointment to inspect the paper files on site. Even being able to find a copy of the *Federal Register* to read the rulemaking proposal required sophistication and resources that many members of the public did not have.<sup>3</sup> Such logistical barriers weeded out many from intelligently participating in the rulemaking process.

Today, because of several statutes and executive branch initiatives, nearly all agency primary rulemaking documents (e.g., notices of proposed rulemaking and final rules) are online.<sup>4</sup> A member of the public interested in viewing proposed and final rule documents can go to Federalregister.gov. To submit a comment on a proposed rule, view other comments, and read supporting materials, a member of the public can go to Regulations.gov, either directly or by a link from Federalregister.gov. Online accessibility therefore has the potential to allow more people to participate in the rulemaking process in an informed, intelligent way. To some extent,

<sup>&</sup>lt;sup>1</sup> See 5 U.S.C. § 553.

 $<sup>^{2}</sup>$  Of course, even these elements are not enough if the agency's rationale for the proposed rule is not written in language accessible to the relevant audience.

<sup>&</sup>lt;sup>3</sup> Cary Coglianese, *E-Rulemaking: Information Technology and the Regulatory Process*, 56 ADMIN. L. REV. 353, 362 (2004) (discussing how access to the *Federal Register* and Code of Federal Regulations was limited).

<sup>&</sup>lt;sup>4</sup> See id. at 363–66 (detailing administrative efforts starting with the Clinton Administration's call to federal agencies to increase the "use of IT in developing and implementing regulations"). See generally Cynthia R. Farina, Reporter, Achieving the Potential: The Future of Federal E-Rulemaking, Report of the Committee on the Status and Future of Federal e-Rulemaking, 62 ADMIN. L. REV. 279 (2010) (reporting on the development of and potential in eRulemaking).

it has advanced that goal.<sup>5</sup> However, it has yet to fulfill its potential, for a variety of reasons that will be discussed in this report.

Part I of this report discusses the process that federal agencies use to conduct rulemaking online. Part II discusses how Regulations.gov/the Federal Docket Management System (FDMS) is governed and funded. Part III discusses how it came into being. Part IV discusses the specific legal requirements that it helps agencies fulfill. Part V examines some of its key problems. Finally, Part VI offers some solutions.

#### Part I. What FDMS/Regulations.gov Looks Like and Its Core Functionalities

Any member of the public can go to Regulations.gov and, once there, search for rulemaking materials, including notices, public comments, and supporting materials. There are several ways that visitors can search for these materials. First, they can enter terms into a search box, just as one would do if searching for materials on a search engine, such as Google. Often, however, this basic search function will yield too many results for visitors to easily sift through to find the desired material. Regulations.gov therefore permits users to perform an "Advanced Search," which allows them to narrow the results by searching fields such as "Document Type" (e.g., "Notice," "Proposed Rule," or "Other"), "Agency," and "Docket Type" ("Rulemaking" or "Nonrulemaking.")

To submit a comment, visitors click on a "Comment Now" button that appears next to a notice. Doing so pulls up a text box with a maximum character count of 5,000 and an option to include attachments. People who wish to submit comments that exceed the character count can type "See Attached" (or similar language) in that text box and upload a longer comment. Depending on the agency, there may be fields within the comment page that require visitors to enter certain information about themselves, such as first name, last name, and contact information (e.g., city, state, and country). Some agencies only ask for this information if a commenter indicates that he or she is submitting a comment on behalf of another.

FDMS is where agency officials create the electronic rulemaking dockets (e-dockets), designated elements of which are viewable on Regulations.gov. An "e-docket" is simply a virtual folder that contains materials relevant to a particular rulemaking, including the notice, supporting materials, and comments. Regulations.gov is the public-facing website that allows people to access materials in the e-dockets. Agencies create and manage the e-dockets and their contents through FDMS.gov, a password-protected site that can be accessed only by authorized agency personnel.

*All* rulemaking notices that are published in the *Federal Register automatically* appear on Regulations.gov. This includes materials from agencies that do not participate in Regulations.gov, such as the Federal Communications Commission (FCC) and the Securities and

<sup>&</sup>lt;sup>5</sup> Congress has also taken a strong interest in advancing this goal. The recently-introduced GOOD Act (or "Guidance Out of Darkness Act") would require agencies to post guidance documents on Regulations.gov and to provide hyperlinks to the posting on the agency website. *See* S. 2296, 115th Cong. (2018).

Exchange Commission (SEC). This is because there is an automated, "behind the scenes" link between Federalregister.gov and FDMS whereby each day, all of the contents of Federalregister.gov are sent to FDMS. Because the FCC and the SEC (and all other nonparticipating agencies) submit their rulemaking materials to the Federal Register, their materials are published on Federalregister.gov and are sent to FDMS, where they then become publicly visible on Regulations.gov.

For the majority of rulemaking agencies that participate in FDMS, once a rulemaking notice arrives in FDMS, an agency user will assign it to an e-docket.<sup>6</sup> With respect to supporting materials, such as Regulatory Flexibility Act analyses, studies, or cases that informed a rule's development but that do not appear on Federalregister.gov, agency officials must first manually upload these to FDMS and then associate them with an e-docket. Once they associate them with an e-docket, they must then designate such items as publicly viewable in order for them to appear on Regulations.gov. Furthermore, assigning documents to a docket allows rulemaking materials to be associated with one another. This association is important for enabling users to find a rulemaking document. It also allows Regulations.gov to include key documents in the history of the rulemaking.

Likewise, comments submitted through Regulations.gov do not automatically appear on Regulations.gov (except if an agency requests this, which very few do). Rather, comments are added to the docket queue on FDMS and an agency official must affirmatively decide whether to associate them with an e-docket, thereby rendering them publicly viewable on Regulations.gov.<sup>7</sup> This gives agencies the chance to review comments for, among other characteristics, profanity, spam, confidential business information, and personally identifiable information.

#### Part II. How FDMS/Regulations.gov is Governed and Funded

FDMS/Regulations.gov is governed by an Executive Steering Committee (Committee) that consists of officials from dozens of federal agencies. The Committee is co-chaired by the Deputy Administrator of the Office of Information and Regulatory Affairs (OIRA) and the Chief Information Officer (CIO) of the Environmental Protection Agency (EPA). It makes decisions about the design, operations, maintenance, and budgeting of FDMS/Regulations.gov upon advice in these areas from several smaller, lower-tiered bodies. These bodies include a Change Control Board, an Advisory Board, and a Budget Working Group.

EPA is considered the "managing partner" of FDMS/Regulations.gov. As such, it is responsible for implementing changes to the system that have been approved by the Committee. To facilitate this responsibility, the EPA created a Project Management Office (PMO), which consists of a small staff of experts in online docket management technology.

There is no direct appropriated funding for FDMS/Regulations.gov. Rather, the system is funded through what eRulemaking officials term a "shared services model." Agencies that participate in eRulemaking fund the system through contributions, decided by a formula. The

<sup>&</sup>lt;sup>6</sup> The independent commissions that have their own eRulemaking systems do not create dockets in FDMS.

<sup>&</sup>lt;sup>7</sup> Agency officials have the option to upload a large batch of comments.

formula for contributions is based, in part, on: 1) the size of the agencies' budgets; 2) the average annual number of rules and non-rule items that agencies publish; and 3) the average annual number of comments agencies receive.<sup>8</sup>

#### Part III. How FDMS/Regulations.gov Came Into Being

In July 2001, President George W. Bush identified "expansion of eGovernment" as one of five priorities of the President's Management Agenda.<sup>9</sup> To support this priority, the Office of Management and Budget (OMB) put in place an implementation strategy for eRulemaking.<sup>10</sup>

In May 2002, the OMB Director issued a memorandum to the heads of agencies stating that OMB would "consolidate redundant IT systems related to rulemaking." OMB initially named the U.S. Department of Transportation (DOT), and then the EPA, the lead agency for this initiative. Both agencies had been operating sophisticated online rulemaking systems for years before FDMS/Regulations.gov was put into place.<sup>11</sup>

Regulations.gov was launched in January 2003.<sup>12</sup> At that time, the public was able to view rulemaking materials available for comment and to submit comments. However, the rulemaking dockets themselves (along with, for example, supporting material and public comments) were not available. In September 2003, EPA, as managing partner of the eRulemaking Program, awarded a contract to Lockheed Martin to integrate various online rulemaking systems with the Regulations.gov portal. EPA and OMB considered three general designs for a new government-wide rulemaking docketing system. The first was a single, centralized system that would replace all existing agency online docketing systems. The second would have allowed agencies with existing online rulemaking dockets to keep those dockets, but they would be linked to a main system used by agencies without their own dockets. The final plan was a "tiered system," which was a hybrid of the two models above.<sup>13</sup>

In February 2004, the Executive Steering Committee, by a vote of 15-2, decided to adopt the first approach.<sup>14</sup> This decision was based on perceived cost savings and ease of searching a centralized system compared to the other two options.<sup>15</sup> In May 2005, agencies began migrating their rulemaking dockets to FDMS. In September 2005, Regulations.gov was updated to allow the public to search and access rulemaking docket contents (e.g., supporting materials and public comments).<sup>16</sup>

<sup>&</sup>lt;sup>8</sup> Curtis Copeland, Cong. Research Serv., RL34210, Electronic Rulemaking in the Federal Government 17 (2008).

<sup>&</sup>lt;sup>9</sup> U.S. GOV'T ACCOUNTABILITY OFFICE, GAO-05-777, PROGRESS MADE IN DEVELOPING CENTRALIZED E-RULEMAKING SYSTEM 6 (2005).

<sup>&</sup>lt;sup>10</sup> See Farina, supra note 4, at 280.

<sup>&</sup>lt;sup>11</sup> Id.

<sup>&</sup>lt;sup>12</sup> Id. <sup>13</sup> Id.

<sup>&</sup>lt;sup>14</sup> COPELAND, *supra* note 8, at 12–13.

<sup>&</sup>lt;sup>15</sup> *Id.* at 13.

<sup>&</sup>lt;sup>16</sup> *Id*. at 14.

By July 2006, seven agencies had migrated to FDMS, and over the ensuing years, more and more agencies did so. Today, all cabinet-level and freestanding Executive Branch agencies (e.g., the EPA) are considered "Participating Agencies" of FDMS/Regulations.gov. This means they maintain their rulemaking and, in some cases, non-rulemaking (e.g., adjudication) dockets on Regulations.gov, and accept comments through the comments feature on the website. To date, there are 184 such "Participating Agencies." Several independent regulatory agencies, such as the Bureau of Consumer Financial Protection and the Federal Trade Commission, are also "Participating Agencies." However, most independent regulatory agencies, such as the FCC, do not participate. Although their rulemaking materials do appear on Regulations.gov, due to the link between Federalregister.gov and FDMS, comments submitted to them do not appear on Regulations.gov, and they do not have e-dockets on Regulations.gov. Several of these nonparticipating agencies maintain their own websites where the public can access a rulemaking docket and comment on a proposed rule.

## Part IV. Participation in FDMS/Regulations.gov Facilitates Compliance with the E-Government Act of 2002

Under the E-Government Act of 2002, agencies must, "[t]o the extent practicable . . . accept submissions under section 553(c) of title 5, United States Code [written data, views, or arguments from interested persons], by electronic means."<sup>17</sup> Furthermore, agencies must "[t]o the extent practicable, as determined by the agency in consultation with the Director [of OMB] . . . ensure that a publicly accessible Federal Government website contains electronic dockets for rulemakings under section 553 of title 5, United States Code ["informal rulemaking"]."<sup>18</sup>

These electronic dockets "shall make publicly available online to the extent practicable, as determined by the agency in consultation with the [OMB] Director . . . all submissions under section 553(c) of title 5, United States Code; and other materials that by agency rule or practice are included in the rulemaking docket under section 553(c) of title 5, United States Code, whether or not submitted electronically."<sup>19</sup>

Although the statute does not require that agencies participate in FDMS/Regulations.gov, participation in FDMS/Regulations.gov allows agencies to fulfill their obligations under the statute. First, when an agency participates in FDMS/Regulations.gov, it maintains an e-docket on FDMS, which appears publicly on Regulations.gov. This fulfills its obligation to "ensure that a publicly accessible Federal Government website contains electronic dockets for . . . [informal rulemaking]."<sup>20</sup> When an e-docket in FDMS/Regulations.gov contains and makes publicly accessible all the materials that the agency normally includes in the rulemaking docket for that rulemaking, and contains public submissions, the docket requirement component of the statute is satisfied. Finally, when agencies participate in FDMS/Regulations.gov, it means they accept

<sup>&</sup>lt;sup>17</sup> E-Government Act of 2002, Pub. L. No. 107-347, § 206(c), 116 Stat. 2899, 2916 (included as a note in 44 U.S.C. § 3501 (2018)).

<sup>&</sup>lt;sup>18</sup> *Id.* § 206(d)(1), 116 Stat. at 2916.

<sup>&</sup>lt;sup>19</sup> *Id.* § 206(d)(2)(B), 116 Stat. at 2916.

 $<sup>^{20}</sup>$  *Id*.

comments from the public, thus fulfilling their obligations to "accept submissions under section 553(c) of title 5, United States Code, by electronic means."<sup>21</sup>

#### Part V. The Key Challenges Underlying Regulations.gov

Before delving into the challenges underlying Regulations.gov, it is useful to first take a step back and keep in mind the extremely tedious work that eRulemaking officials completed to make online rulemaking a reality. The magnitude of this achievement cannot be overstated. Achieving the goal of centralized online rulemaking required tackling an extremely complex information management problem, and eRulemaking officials have managed to make important rulemaking information substantially more accessible to the public than in the era before the existence of FDMS/Regulations.gov.

The limitations of the website, which will be explored below, are not in any way a criticism of the dedication of eRulemaking officials. They are, rather, a byproduct of a system designed to store an amazing variety and quantity of rulemaking materials from nearly 200 agencies (and counting) in one place, and one in which all of these agencies are responsible for their own submissions and docket management.<sup>22</sup> However, with the appropriate resources and prioritization, the challenges can be overcome.

To understand the main challenges with FDMS/Regulations.gov, it is helpful to again consider its purpose. It was created to make it easier for the public to participate in the noticeand-comment process in an informed way and to readily access materials that reveal the status and outcome of the rulemaking process. The eRulemaking Program envisioned that it would do so by allowing the public to 1) access the text of the Notice of Proposed Rulemaking (NPRM) and, if existent, final rule and accompanying explanation; 2) access materials upon which the agency substantially relied to develop the proposed rule; 3) submit comments online and review the comments others have submitted; and 4) follow the course of a rulemaking to determine whether the NPRM has been supplemented, finalized, withdrawn, etc. However, many users of Regulations.gov have found that the system does not allow people to consistently and reliably: a) search for and find all documents related to a particular rulemaking and b) access supporting materials and other relevant information about rulemakings, for reasons that will be discussed below.<sup>23</sup>

To uncover some of the major flaws with Regulations.gov, I extensively used the site to attempt to find rulemaking materials and consulted with academics who have written on the

<sup>&</sup>lt;sup>21</sup> Id.

<sup>&</sup>lt;sup>22</sup> Organizations wishing to engage in comprehensive analyses of public participation in rulemaking are hampered by the fact that although most agencies participate in Regulations.gov, several do not, making it an onerous process to include those agencies' data. *See, e.g.,* apendleton, *Regulations.gov Continues to Improve, but Still Has Potential for Growth,* The Sunlight Foundation (Apr. 9, 2013, 11:21 AM),

https://sunlightfoundation.com/2013/04/09/regulations-gov-continues-to-improve-but-still-has-potential-for-growth/. <sup>23</sup> Another problem that non-agency users have pointed out is that a user cannot reliably determine how many

comments were submitted on a rulemaking and whether the comments visible on Regulations.gov are all the comments the agency received.

subject. I also had numerous discussions with members of the public who use Regulations.gov regularly.

#### A. <u>Users Find It is Difficult to Consistently and Reliably Search for Rulemaking</u> <u>Materials</u>

One reason it is difficult to reliably and consistently find rulemaking materials is because agencies sometimes create multiple e-dockets for the same rulemaking. For example, if an agency's rulemaking has gone from an NPRM to a final rule, the agency sometimes creates a separate e-docket for the final rule, instead of maintaining a single e-docket to which all documents related to the rulemaking are assigned. A user who tries to find this rule might come across the first e-docket the agency created and conclude incorrectly that there has been no final rule issued.<sup>24</sup> Sometimes, this "multiple e-docket" problem happens because a sub agency (e.g., the Occupational Safety and Health Administration) issued the NPRM and created the initial docket, and the parent agency (e.g., Department of Labor) issued the final rule and created the second docket. In any case, in many instances, there are at least two e-dockets, each containing documents that are part of a single rulemaking.<sup>25</sup> At best, this is confusing. At worst, it misleads the user as to the status of the rulemaking if her search does not locate both dockets and enable her to recognize the relationship between them.

Another reason it is difficult to search for rulemaking materials is because the "Advanced Search" feature on Regulations.gov in many instances does not helpfully narrow down the number of results that come up in a search. The purpose of an "Advanced Search" is to allow users to search by different filters (e.g., date range, type of source, author, and so on), reduce the number of search results, and therefore increase the likelihood of finding what the user is looking for.

For example, suppose someone would like to use Google to find an article she read about robotics, and she recalls that she read the article in 2006. If she were to search Google for this particular article without using an advanced search, she would likely have to sift through millions of results before she came across what she was looking for. However, if she were to use Google's "Advanced Search" feature, she could select the date range as "1/1/2006-12/31/2006," thereby drastically decreasing the number of results that come up and increasing the likelihood that she will find the relevant article.

<sup>&</sup>lt;sup>24</sup> Other agency practices sometimes compound the problem of finding all documents related to a rulemaking. In many instances, the title of the final rule does not match the title of the proposed rule, especially if time has passed and there have been, for example, supplemental NPRMs or other shifts in the focus of the rule. Sometimes agencies "reuse" titles, so that it becomes difficult to identify the documents for which the user is looking.

<sup>&</sup>lt;sup>25</sup> Each e-docket has a unique docket number assigned by FDMS. Docket numbers begin with an alphabetic prefix unique to the agency. So, for example, OSHA's dockets begin "OSHA" and the Department of Labor's dockets begin "DOL." The remaining components of a typical e-docket number are a four-digit date and a three or four-digit number that is the docket number. Each document in the docket is identified by a *document number* that is the docket number plus an additional three (or more) digit number that is the sequential order in which that document was added to the docket. Hence, the document number is the unique identifier of each document and integrally links it with its home docket and, in turn, all the other documents in that docket.

In contrast, many of the filters that appear on Regulations.gov's "Advanced Search" feature do not helpfully narrow down relevant results. One of the first search filters that a user encounters on the Advanced Search page is "Document Type." The options presented here are: "Notice"; "Proposed Rule"; "Rule"; "Supporting & Related Materials"; "Other"; and "Public Submission." One problem presented by these options is that they are not mutually exclusive. A "Proposed Rule" and a "Rule" are both "Notices." If I am interested in commenting on a particular NPRM, and I go to this Advanced Search page to find that NPRM, it is not clear whether I should select "Notice" or if I should select "Proposed Rule." Similarly, if I am interested in commenting on a particular advanced notice of proposed rulemaking (ANPRM), it is not clear whether I should select "Notice," "Proposed Rule," or "Other."<sup>26</sup>

The second problem presented by this filter is that it is not comprehensive. Section 553 of Title 5, U.S. Code (Section 553) governs informal rulemaking, but it establishes only minimal procedural requirements, thereby effectively obscuring the complexity of the rulemaking process.<sup>27</sup> Looking only at Section 553, one might think that agencies publish only two documents during the course of a rulemaking: an NPRM and a final rule.<sup>28</sup> In reality, however, agencies publish a remarkable variety of documents during the course of a rulemaking, revealing a more complex and nuanced process than that which is suggested by Section 553.<sup>30</sup>

For example, agencies may engage the public before proposing a rule by issuing an ANPRM or a "notice of inquiry."<sup>31</sup> They may propose rules by publishing a "notice of intent to grant a rulemaking petition" or seek additional public input through a "supplemental notice of proposed rulemaking" or a "notice of extension of the comment period."<sup>32</sup> Agencies may choose, or be statutorily required, to hold public hearings on the proposed rule, and notices of these hearings, agendas, etc. often appear on Regulations.gov.<sup>33</sup> The rulemaking may generate a Paperwork Reduction Act submission that the agency notices and takes comments on. Moreover, agencies routinely promulgate a variety of rules in addition to final rules, including "temporary rules," "interim final rules," and "direct final rules," among others. All of these documents—and many more besides—fit within the basic structure of informal rulemaking as set forth in Section 553 of Title 5.<sup>34</sup>

<sup>&</sup>lt;sup>26</sup> It is worth emphasizing that this confusion is not confined to the public user. At the time that documents are uploaded to FDMS and assigned to dockets, the agency must supply appropriate categorizations. Hence, the person doing data entry must also decide if, for example, an ANPRM or a notice of extension of the comment period should be categorized as a "Proposed Rule," a "Notice," or "Other." Understandably, inconsistent categorization often occurs, within the same agency as well as across agencies.

<sup>&</sup>lt;sup>27</sup> Memorandum from Emily Bremer 2 (Mar. 14, 2013) (on file with author).

<sup>&</sup>lt;sup>28</sup> Id.

<sup>&</sup>lt;sup>29</sup> Id.

<sup>&</sup>lt;sup>30</sup> Id.

 $<sup>^{31}</sup>$  *Id*.

<sup>&</sup>lt;sup>32</sup> Id. <sup>33</sup> Id.

<sup>&</sup>lt;sup>33</sup> Id. <sup>34</sup> Id.

Compounding this complexity is the issue of guidance.<sup>35</sup> Section 553 exempts interpretive rules, guidance, policy statements, and other documents from notice-and-comment requirements. Such documents may thus be viewed as technically not part of the rulemaking process encompassed by FDMS/Regulations.gov. On the other hand, agencies often voluntarily elect to craft guidance documents through notice and comment and in some instances are required to seek comment under OMB's Good Guidance Practices directive. There is great variety in the terminology agencies use to describe these documents. The upshot of this extraordinary complexity is that the list of "Document Type" options, as currently structured, makes it difficult for the visitor to appreciate just what, exactly, is out there.<sup>36</sup>

In addition to the foregoing problems presented by the "Document Type" filter, agencies do not use the "Document Type" categories in a consistent manner. For example, some agencies, when uploading documents to FDMS, tag an ANPRM as a "Proposed Rule." Others tag it as a "Notice." And still others tag it "Other." Some agencies, when uploading a notice of public meeting within a rulemaking, tag the document as a "Proposed Rule," and some tag it as a "Notice." Even within a single agency, inconsistency may occur in the use of these categories.

For these reasons, in order to derive any benefit from using the "Document Type" filter, a visitor, at a minimum, must be aware of the particular methods of "Document Type" tagging from the agency of interest. Even then, the visitor cannot reliably select a single "Document Type" to aid their search, due to the possibility of inconsistent categorization at the point of data entry. The "Document Type" filter is therefore often not helpful in narrowing search results and may create the misleading impression that the document sought does not exist.

In addition to the "Document Type" filter (for which, recall, the options are "Notice;" "Proposed Rule;" "Rule;" "Supporting & Related Materials;" "Other;" and "Public Submission"), there is a "Document Subtype" filter. This filter can only be used if the user has selected an "Agency" (the "Agency" filter will be discussed below). As is the case with "Document Type," when an agency uploads a document to FDMS (or associates a document with a docket), the data entry person may identify the document as belonging to a particular "Document Subtype." However, unlike with "Document Types," agencies are not required to designate a "Subtype." If an agency elects to use "Document Subtypes," however, the agency's FDMS Agency Administrator configures a standardized dropdown menu of subtypes that are available to that agency's FDMS users. The subtypes available for inclusion in an agency's dropdown menu must be selected by the FDMS Agency Administrator from a finite list of shared, standardized subtypes maintained by the eRulemaking PMO. If an agency requires a new subtype, it must first submit the subtype to the eRulemaking PMO for approval. Examples of "Document Subtypes" are: "Advanced Notice of Proposed Rulemaking," "NEPA Documentation," "Final Rule," and "Supplemental Notice of Proposed Rulemaking."

<sup>&</sup>lt;sup>35</sup> Under the Administrative Procedure Act, a specific kind of guidance document, called an "interpretative rule," is considered a rule. *See* 5 U.S.C. § 553(b)(3)(A).

<sup>&</sup>lt;sup>36</sup> The GOOD Act, currently under consideration by the U.S. Senate, would require agencies to publish guidance documents on Regulations.gov and provide a hyperlink to the publication on their own. The bill defines "guidance document" broadly to include such things as blog posts, news releases, etc. *See* S. 2296, 115th Cong. (2018).

Although the eRulemaking PMO's role in the process provides some measure of uniformity, agency discretion to opt out of using "Document Subtype" designations, draft agency-specific "Docket Subtype" menus, and request the creation of new "Document Subtypes" appears to have resulted in significant variation among agency practices and contributed to an overwhelming number of "Document Subtypes" in use on Regulations.gov.<sup>37</sup> Because it offers greater levels of specificity, the "Document Subtype" filter can potentially help solve the non-comprehensiveness problem discussed above with respect to "Document Type." However, inconsistent practices with respect to its use undermines its effectiveness.

The next filter that a visitor encounters on the "Advanced Search" page is "In these date ranges." The options here are: "Comment Due Date;" "Comment Start Date;" "Creation Date;" "Posted Date;" and "Received Date." The first two are fairly straightforward: they simply refer to the date that comments are due, and the dates that the agency first accepted comments on the rulemaking. The last three are not intuitive. While these terms may make perfect sense to the agency at the point a document is uploaded or a comment is released for public visibility, the public user is unlikely to understand the difference between "Creation," "Posted," and "Received," and the page does not define these terms.

Immediately below this filter, the user can search by "Agency." This seems fairly straightforward upon first glance. However, as with many of the other filters, visitors who use this may be misled. The problem here is that selecting a parent agency as the "Agency" does not include results for sub agencies. For example, suppose visitors are interested in reading and submitting a comment on the recently posted "Request for Comments on the Cross-Agency Priority Goal: Leveraging Data as a Strategic Asset." Perhaps they came across that document on the *Federal Register* website. If they did, they would see that the agency that issued this particular notice is listed as "Department of Commerce." When the prospective commenters go to Regulations.gov to try to find and comment on the document, they would, quite sensibly, select "Department of Commerce" as the "Agency" that created the docket in FDMS was, in this instance, a sub agency of the Department of Commerce: the Bureau of the Census. Searching by "Department of Commerce" will not pull up the document. Visitors who use this filter and select a parent agency may erroneously conclude that a particular document has not been published.

One of the next filters on the "Advanced Search" page is "Docket Type." Here, the user may select "Rulemaking" and "Nonrulemaking." Given that the name of the website is "Regulations.gov," and "regulation" is often used synonymously with "rulemaking," the appearance of "Nonrulemaking" as a "Docket Type" may cause confusion. Some agencies include non-binding guidance materials under the "Nonrulemaking" "Docket Type." Some agencies, such as the U.S. Department of Transportation, include their adjudication dockets under "Nonrulemaking." Regulations.gov does not define the term "Nonrulemaking," but this would be very helpful information for a member of the public to know before searching.

<sup>&</sup>lt;sup>37</sup> See Bremer, supra note 27, at 5.

There are additional complexities that one will encounter if one decides to use the "Docket Type" filter. For example, a visitor who selects "Nonrulemaking" as the "Docket Type," and "Proposed Rule" and "Rule" as "Document Types," will get 23,328 results. On its face, it is difficult to comprehend how it can be the case that there are any instances in which a "Proposed Rule" or a "Rule" can be part of a "Nonrulemaking" docket. It is possible that some agencies' adjudication dockets contain "Proposed Rules" and "Rules" as "Supporting and Related Materials." That would be a sensible explanation for a "Proposed Rule" and a "Rule" being part of a "Nonrulemaking docket." However, it appears that in most instances in which a "Proposed Rule" and a "Rule" are part of a "Nonrulemaking Docket," the "Proposed Rule" or "Rule" is the primary document. This means that either the agency official incorrectly labeled the docket "Nonrulemaking," or incorrectly labeled, say, an adjudication order (or other kind of "Nonrulemaking document") as a "Proposed Rule" or a "Rule."

An additional problem with the "Docket Type" filter is that if a user selects *either* "Rulemaking" or "Nonrulemaking" as the "Docket Type," Regulations.gov displays materials that are not associated with *any* docket at all (e.g., documents from agencies that do not participate in FDMS/Regulations.gov). It could be confusing for a member of the public to see a freestanding document (say, an order from the SEC) to come up in a search of "Docket Types."

In addition to the "Docket Type" filter, there is a "Docket Subtype," as well as a "Docket Subtype Level 2" filter. The "Docket Subtype" filter can only be used if the visitor has selected an "Agency," and the "Docket Subtype Level 2" filter can only be used if the visitor has selected an "Agency" *and* a "Docket Subtype." As is the case with "Document Subtype," agency officials *can* identify dockets as belonging to a particular "Docket Subtype" and "Docket Subtype Level 2," but they are *not required* to use these designations. Most agencies use the "Docket Subtype" without using "Docket Subtype Level 2." As with "Document Subtypes," if an agency elects to use "Docket Subtype" and "Docket Subtype Level 2." the agency's FDMS administrator configures a standardized dropdown menu of subtypes that will be available to that agency's FDMS users. Examples of "Docket Subtypes" are: "Consultations Rulemaking," "Statistics," and "Health Rulemaking." Examples of "Docket Subtypes Level 2" are: "101- Balloons, Kites, Rockets/Free Balloons," "Ultralight Vehicles," and "Airport Security."

As is the case with "Document Subtypes," agency discretion to opt out of using "Docket Subtype" and "Docket Subtype Level 2" designations, draft agency-specific subtype menus, and request the creation of new subtypes appears to have resulted in significant variation among agency practices and contributed to an overwhelming number of "Docket Subtypes" and "Docket Subtypes Level 2" in use on Regulations.gov. This results in the same problems discussed above with respect to Document Subtypes and therefore further hinders the ability of visitors to use "Advanced Search" to find relevant results.

#### B. E-Dockets Are Not Reliably Informative

Many e-dockets on Regulations.gov do not contain all relevant Unified Agenda<sup>38</sup> information (e.g. whether it was published in the fall or spring edition; whether it is a "Major Rule," and whether there are "federalism implications"). Others do include this information. The piece of information that allows FDMS to link a rulemaking e-docket to the appropriate Unified Agenda entry is the Regulation Identifier Number (RIN). In some instances, the absence of Unified Agenda information may indicate that the rulemaking was not included in the Unified Agenda, but more frequently the problem is that agencies are not required to enter a RIN when a new rulemaking e-docket is created—and do not do so. Executive Order 12,866 requires that all regulatory actions have a RIN.<sup>39</sup> In practice, a RIN is generated when an agency submits information for a new rulemaking action to the Regulatory Information Services Center (RISC) for inclusion in the Unified Agenda. The RIN is eight digits, the first four of which are unique to the agency, and the second four of which are unique to the particular rulemaking action. When an agency official enters a RIN for a docket in FDMS, the Unified Agenda information is automatically included in the e-docket displayed on Regulations.gov. This occurs because there is a behind the scenes link between FDMS and the Unified Agenda.

In some instances, a RIN may not be in existence at the time the e-docket is created. However, the far more common problem appears to be simply the failure to enter a RIN in the appropriate field at the time the e-docket is created—or to amend the docket description at the time when a document having a RIN is added. In many instances when a RIN is not displayed on Regulations.gov (and hence no Unified Agenda information is linked), the RIN can be found in the online *Federal Register* version of the document and/or in the text of the document itself. Extremely inconsistent agency practice with respect to providing the RIN exacerbates search problems: users who are sophisticated enough to try to search by RIN (which, for example, they found in the *Federal Register* notice) may not find documents that are in fact part of the relevant rulemaking.<sup>40</sup>

Yet another problem with respect to e-dockets is they do not always contain supporting materials that are visible to the public. If a visitor opens an e-docket that does not have supporting material, the visitor would see, under "Supporting Documents," "No documents available." As discussed above, there are good, practical reasons for agencies to include supporting materials within their e-dockets. Doing so helps boost the quality of public comments. It might be consistent with the e-Government Act for the agency not to include any supporting materials (the e-Government Act, after all, only requires inclusion of these materials if "practicable" and if the agency, by rule or practice, includes them in their rulemaking dockets). However, leaving the "Supporting Documents" section of the e-docket without materials may diminish the public's ability to adequately comment.

<sup>&</sup>lt;sup>38</sup> The Unified Agenda is a semi-annual publication of significant regulatory actions that agencies plan to take in the short and long term. The Unified Agenda requires agencies to indicate, among other things, whether a rule has federalism implications, creates unfunded mandates, or affects small entities.

<sup>&</sup>lt;sup>39</sup> See Exec. Order No. 12,866, 58 Fed. Reg. 51,735 (Sept. 30, 1993) (Section 4(b)).

<sup>&</sup>lt;sup>40</sup> Additional problems exist with RINs, including the use of a single RIN for multiple rulemakings and the assignment of multiple RINs to a single rulemaking.

#### **Part VI. Solutions**

Improving Regulations.gov/FDMS calls for a combination of changes to the system and changes to individual agency practices in using the system. Of course, the system's design and agency practices in using the system affect each other synergistically.

As a first step, the eRulemaking Program should work with the Office of the Chairman of the Administrative conference to provide, on an ongoing basis, resources to help identify and meet user needs in navigating and finding materials on Regulations.gov, both in its current form and as it continues to evolve. As part of this process, it should implement a method for allowing users easily to find definitions of important and potentially confusing terms used in the search process. It should then consider implementing the changes suggested below.

#### Problem: Agencies Sometimes Create Separate E-Dockets for the Same Rulemaking

#### Solution:

The eRulemaking Program should configure FDMS/Regulations.gov so that it warns an agency user of FDMS and a public user of Regulations.gov when it appears that there are multiple dockets for the same rulemaking. When multiple e-dockets are detected, a warning message should be sent to the relevant agencies on FDMS. It would ultimately be up to the agencies to decide how to respond to the message, though agencies should take them seriously and strongly consider merging all dockets that pertain to a single rulemaking. A warning message should also appear on Regulations.gov, informing the public that another e-docket may contain relevant information. The other e-docket should be identified by docket number, name, and any other relevant identifying information so the public can easily access that e-docket and determine whether there is, in fact, any relevant information.

#### Problem: The "Advanced Search" Feature Creates Complexities for Users of Regulations.gov

#### Solution:

The eRulemaking Program should consider enlisting government and private sector experts in rulemaking, information technology, data management, and organization to work closely with one another at all stages of whatever process is ultimately adopted to help improve the "Advanced Search" feature. The process suggested below could be among those that this group of experts considers using. Whether the group ends up adopting the process below, or some other process, it will likely be a large, resource-intensive process. No single agency or office (e.g., the PMO) alone should be expected to undertake it.

First, the "Notice" option under "Document Type" should be marked for elimination so that the remaining options are mutually exclusive. "Other" should also be marked for elimination. Then, all of the "Document Subtypes," "Docket Subtypes," and "Docket Subtypes Level 2" should be laid out and organized by agency. Any option that includes another (e.g., "Notice" includes "Advanced Notice of Proposed Rulemaking") should be marked for elimination so that all of the subtype options within an agency are mutually exclusive.

Additionally, "catch all" categories such as "Other" and "Miscellaneous" should be marked for elimination.

Next, a random, sufficiently large sample of all documents on Regulations.gov should be generated. Each one of these documents should be carefully analyzed to determine, among other factors, the nature of the underlying agency process that led to the document, the role the document played in the process, the public response sought by the agency (if any), and the practical consequences that might flow from the document's publication.

Based on the analysis of these documents, the "Document" and "Docket" "Type" and "Subtype" options should be revised. The revised options should sufficiently capture the diversity of documents available on Regulations.gov, should be mutually exclusive, and should be specific (i.e., there should be no open-ended options such as "Miscellaneous"). Additionally, the set of options should be manageable in number and understandable to the general public (with accompanying guidance material). Each of the revised options should include several specific examples. Additionally, depending on the results, there should be openness to the possibility of marking for elimination the "Subtype" filters entirely.

In selecting and analyzing the different kinds of materials on Regulations.gov, the eRulemaking Program can draw upon the research of Professor Emily Bremer. Professor Bremer, when she was an attorney at the Conference, undertook a comprehensive examination of this sort, and created categories, which could serve as a prototype for a revised set of "Document Types" and "Document Subtypes." The Appendix displays an excerpt of Professor Bremer's work.

After a revised classification scheme for document and docket types and subtypes is settled upon, the public should be given the opportunity to comment on the proposed new scheme, and changes should be considered based on these comments. The Executive Steering Committee should then vote on the scheme. If it is adopted, the eRulemaking Program should publish it on Regulations.gov and widely disseminate it to agency officials. FDMS should then be reconfigured to present this approved list of tags as the choices from which agencies may select when they create e-dockets or upload documents. The eRulemaking Program should offer training on how documents and dockets are to be classified under the new scheme.

There are at least two approaches the eRulemaking Program can take with respect to the mechanics of how documents are to be classified under the new scheme. Under one approach, agency officials would be responsible for selecting types or subtypes within FDMS, just as they do now. There would be some questions posed to the person entering the data to help ensure that she applies the correct categorization (e.g., "What is the purpose of this document? Is it to solicit input on a rulemaking proposal? If so, consider using one of these labels: [x], [y], [x].") Alternatively, the eRulemaking Program could, subject to available technology, incorporate the revised scheme into FDMS's decision logic. Under this approach, agency officials would no longer be responsible for selecting types or subtypes within FDMS. Rather, after they upload documents to FDMS, or associate a document with a docket (as in the case of a rulemaking document that has been sent to FDMS via the *Federal Register*), FDMS would "read" the

document and automatically tag it with the appropriate type or subtype based on the software's analysis of the document's text. An automated approach of this sort should seriously be considered, given that there are millions of documents on Regulations.gov, many of which need to be reclassified.

Under either approach, FDMS users should have the ability to override an initial classification if they believe it is an error. Furthermore, agency officials and members of the public should be given the opportunity to easily flag for review any documents and dockets they believe the system tagged in a way that is inconsistent with the revised classification scheme.

As this new classification scheme is being rolled out, there should be a way to preserve historical data. Suppose an automated approach is employed to reclassify the millions of existing documents on Regulations.gov. What happens to the original classifications? There might very well be some reason why a member of the public might want to know, say, how many documents were tagged as "Notices" in 2012. If, under the new classification scheme, "Notices" is no longer a category, and there was no preservation of the original classification data, that person would not be able to answer the question. eRulemaking officials should therefore contemplate ways to preserve the original classifications as the new classification scheme is deployed.

eRulemaking officials should also consider how to remedy the errors with respect to the "Docket Type" filter. They should ensure that NPRMs, final rules, and other kinds of rulemaking documents do not appear in a search of "Nonrulemaking" dockets if they are the "Primary Documents" within those dockets. They should also consider ensuring that freestanding documents, such as those from non-participating agencies, do not appear in any search of "Dockets," whether "Rulemaking" or "Nonrulemaking."

The final issue identified above with respect to "Advanced Searching" is the "Search by Agency" filter. Recall that a visitor who searches by parent agency may miss all documents posted by a sub agency. To remedy this problem, Regulations.gov should display a message whenever a visitor searches by an agency for which there are participating sub agencies. The message should ask the searcher whether she has found what she was looking for and, if not, whether she wishes to run the search again "in related agencies" or some similar language.

#### Problem: Unified Agenda Information Does Not Appear Within All E-Dockets

#### Solution:

The eRulemaking Program should ensure that if a RIN appears in the heading of a *Federal Register* page associated with a document, or is identified within the text of a document itself, it is also included within the Regulations.gov e-docket. Recall that including the RIN in the e-docket automatically causes Unified Agenda information to be displayed within the docket. If no RIN is included in the e-docket, no Unified Agenda information is displayed.

To ensure that if a RIN exists, it appears within the relevant Regulations.gov e-docket, the eRulemaking Program should collaborate with GSA's Regulatory Information Services Center to establish common standards for sharing RIN information over the internet. The eRulemaking Program should also consider establishing RIN as a mandatory field in FDMS, unless a docket manager confirms that the docket does not have a corresponding RIN. Docket managers should be presented with a list of available RINs from ROCIS/RegInfo, and make a positive confirmation of which one to use. There should also be an automatic cross-check to identify which RINs have not been assigned to a docket. Through a combination of notifications and workflow actions, the eRulemaking Program should take proactive steps to find these "orphan" RINs a docket "home." For all documents for which no RIN has been generated, FDMS should automatically cause a message to be displayed within the e-docket that states: "Unified Agenda Information Not Available Because No RIN Has Been Generated."

Under this approach, agency officials would still be responsible for including the RIN as they create a docket. The technology would merely serve as a check on agency officials' work so as to reduce human error. As with "Document Type" and "Docket Type" classifications, agency officials and members of the public should be given the opportunity to easily flag any RINs they believe are incorrect, or incorrectly omitted.

#### Problem: Agencies Do Not Always Include Supporting and Related Materials Within E-Dockets

#### Solution:

The eRulemaking Program should ensure that agencies receive prompts that alert them to any dockets that do not have supporting and related materials. The prompt should state something to the effect of: "This docket must by law include, to the extent practicable, all materials that by agency rule or practice are included in the rulemaking docket, whether or not submitted electronically." Agency officials would be responsible for deciding how to respond to the message, if at all.

#### Acknowledgments

I am deeply grateful to Cynthia Farina and Cary Coglianese, both of whom offered expert feedback on this report and who contributed enormously to the project overall. I am also indebted to numerous professionals throughout the federal government and private sector who offered extraordinarily helpful feedback on this report, and with whom I have worked collaboratively on this project. Without their creativity and open-mindedness, this project would not have been possible.

#### Appendix

Note: The following is an excerpt from a memorandum written by Emily Bremer while she was an attorney at the Conference.

As previously explained, I have organized the rulemaking documents identified through my review of document subtypes. It bears noting that for purposes of this discussion, I define "rulemaking documents" as documents used by agencies to conduct the informal rulemaking process. In keeping with this process-based focus, the documents are organized first into four broad categories, based on the stage of the rulemaking process during which the documents are used. These categories include pre-rulemaking, rulemaking, rules, and post-rulemaking. Within these broad, stage-based categories, documents are further categorized based on their specific purpose or role in that stage of the process. For each individual document, the name of the document is provided, along with the identity of the agency or agencies that published the particular example document(s) I relied upon, and (in parentheses) the section of the *Federal Register* in which the example document is published, if at all. This part concludes with a few other observations and analysis of issues identified during my review.

It bears noting that this is probably not an exhaustive list of rulemaking documents. Although I identified most of these documents through my thorough examination of the documents pulled through searches of Regulations.gov and the *Federal Register* using information provided in the subtype data set, and some of document names listed here are also used as names for subtypes that have been approved for use on FDMS, this is not a list of document *subtypes* per se. I anticipate adding to the list as I complete the remaining research and identify additional documents used by agencies as vehicles of the informal rulemaking process.

#### A. Pre-Rulemaking Documents

#### 1. Requests for Information that May Support or Inform a Proposed Rule

- *Advance Notice of Proposed Rulemaking*—Treasury (proposed rules)
- *Notice of Inquiry*—FCC (not published in the *Federal Register*)
- *Request for Information*—Labor ESA (proposed rules; combined with extension of comment period)

#### 2. Requests for Further Pre-Rule Public Input

- *Notice of Hearing*—Labor OSHA (notices; seeking input on regulatory agenda and priorities)
- *Notice of Extension of Comment Period*—Treasury (proposed rules), CPSC (proposed rules)

#### 3. Negotiated Rulemaking Documents

- *Notice of Intent to Form a Negotiated Rulemaking Committee*—Interior (proposed rules; consolidated with Request for Nominations)
- *Notice of Establishment*—HHS (notices)
- *Request for Nominations*—Interior (proposed rules; consolidated with Notice of Intent to Form a Negotiated Rulemaking Committee)

#### **B.** Rulemaking Documents

#### 1. Proposed Rules for Public Comment

- Notice of Proposed Rulemaking—5 U.S.C. § 553(b)
- *Supplemental Notice of Proposed Rulemaking*—EPA (proposed rules; combined with an extension of the comment period)
- Prepublication Posting of Proposed Rules and Other Documents
  - *Prepublication Display*—CMS (refers to prepublication posting on Regulations.gov or the agency's website of a NPRM (or other document) that has been submitted to the *Federal Register* for publication)
  - Signed Federal Register Document—EPA (used for the same purpose as "prepublication display")
- Subject Matter Based Subtypes Used for NPRMs and SNPRMs—For FWS and EPA in particular, the eRulemaking PMO has approved subtype designations based on the subject matter of a proposed rule, which the agency routinely includes in the title of its *Federal Register* documents.<sup>41</sup> In some cases, these subtypes that have been approved for use are based on subjects that are subsidiary to a larger subject. In such cases, both the primary and subsidiary subjects are typically included in the title of document(s) published in the *Federal Register*.
  - *Migratory Bird Hunting*—Interior FWS (proposed rules; examples are actually SNPRMs)
  - Approval and Promulgation of State Implementation Plans—EPA (proposed rules)
  - Public Hearings and Submission of Plans—EPA (proposed rules)
  - Endangered and Threatened Wildlife and Plants
    - Proposed Critical Habitat Designation—Interior FWS (proposed rules)
    - Proposed Species Listing—Interior FWS (proposed rules)
    - Proposed Species Reclassification—Interior FWS (proposed rules)
    - Proposed Establishment of Nonessential Experimental Population— Interior FWS (proposed rules)

<sup>&</sup>lt;sup>41</sup> The subtype data set included rulemaking subtypes that have been approved for use on FDMS. In some cases, agencies have requested and secured approval for a subtype, but have not configured the subtype for use, and are therefore not actively using the subtype to categorize documents on FDMS.

- *Migratory Bird Subsistence Harvest in Alaska*—Interior FWS (proposed rules)
- *Injurious Wildlife Species*—Interior FWS (proposed rules)
- *Taking and Importing Marine Mammals*—Commerce NOAA (proposed rules)
- *Refuge-Specific Regulations*—Interior FWS (proposed rules)
- *Eagle Permits*—Interior FWS (proposed rules)
- *Migratory Bird Permits*—Interior FWS (proposed rules)
- Petitions for Rulemaking
  - *Notice of Receipt of Petition for Rulemaking*—NRC (proposed rules)
  - Notice of Availability and Request for Comments—DHS Coast Guard (proposed rules)

#### 2. Requests for Further Public Input on Proposed Rules

- *Notice of Extension of Comment Period*—Interior (proposed rules); Defense (rules and regulations)
- *Reopening of Comment Period*—EPA (proposed rules)
- *Request for Comments*—FCC (proposed rules; requesting comments on a new development relevant to an ongoing rulemaking)
- *Notice of Data Availability*—EPA (proposed rules; giving notice that new data relevant to an ongoing a rulemaking is available and providing an opportunity for the public to comment on it)
- *Notice of Public Meetings*—DOT FTA (notices)
  - *Notice of Stakeholder Meeting*—EPA (notes from meeting posted on regs.gov suggest meetings are sometimes used to engage the public on proposed rules).
  - *Notice of Meeting*—NRC (proposed rules)
- *Notice of Public Hearing*—Copyright Office (proposed rules); OSHA (proposed rules, though identified on Regulations.gov as notices); FMCSA (proposed rules; called a "Notice of Change in Hearing Structure," updating previously announced plan for public hearing on proposed rule)

#### 3. Proposed Changes to Promulgated Rules

• *Notice of Proposed Extension of Effective Date*—OPM (proposed rules; combined with proposal to revoke previously published final rule)

#### 4. Non-Promulgation Termination of a Rulemaking Proceeding

- Notice of Termination
  - Notice of Termination—DHS Coast Guard (proposed rules)
  - *Notice of Termination of Proceeding*—DOT (proposed rules)
  - Notice of Termination of Rulemaking—DOT NHTSA (proposed rules)

- Notice of Termination of Proposed Rule
- Withdrawal
  - *Withdrawal*—EPA (proposed rules)
  - *Withdrawal of Proposed Rule*—VA (proposed rules); HUD (proposed rules)
  - Withdrawal of Proposed Rule and Closure of Petition for Rulemaking—NRC (proposed rules)
- Denial of Petition
  - *Denial of Petition for Rulemaking*—NRC (proposed rules)
- Resolution
  - *Resolution and Closure of Petition Docket*—NRC (proposed rules)

#### C. Rules

#### 1. Temporary Rules

- *Temporary Rule*—Commerce NOAA (rules and regulations; also identified as an emergency action)
- *Temporary Interim Rule*—DHS Coast Guard (rules and regulations)

#### 2. Interim Rules

- *Notice of Tentative Final Order*—FDA (older documents; not clear how published in *Federal Register*)
- *Tentative Final Regulation*—FDA (rules and regulations)
- Interim Final Rules
  - Supplemental Interim Final Rule—Commerce ITC (rules and regulations)
  - Interim Final Rule with Request for Comments—HHS (rules and regulations)
- *Emergency Airworthiness Directive*—FAA (rules and regulations)

#### 3. Direct Final Rules

- *Direct Final Rule*—EPA (rules and regulations)
- *Giving Effect to Consent Decree*—EPA (rules and regulations)
- *Airworthiness Directives*—FAA (rules and regulations)
- *State Implementation Plans*—EPA (rules and regulations)

#### 4. Final Rules

- *Prepublication Display*—EPA (rules and regulations; refers to the pre-publication posting on Regulations.gov of a final rule or other document after it has been submitted to the *Federal Register* for publication)
- *Final Rule*—Commerce NOAA (rules and regulations)

- *Final Rule with Request for Comments*—FAA (rules and regulations; airworthiness directive)
- *Affirmation of Interim Rule*—Agriculture APHIS (rules and regulations)
- *Subject Matter Based Subtypes Used for Final Rules*—As with proposed rules, FWS and EPA have secured approval to use subtype designations based on the subject matter of a final rule.
  - Endangered and Threatened Wildlife and Plants
    - Species Listing—Interior FWS (rules and regulations)
    - Establishment of Nonessential Experimental Population—Interior FWS (rules and regulations)
    - Species Reclassification—Interior FWS (rules and regulations)
    - *Species Delisting*—Interior FWS (rules and regulations)
    - *Critical Habitat Designation*—Interior FWS (rules and regulations)
  - Injurious Wildlife Species—Interior FWS (rules and regulations)
  - Incidental Take of Marine Mammals—Commerce NOAA (rules and regulations)
  - *Refuge-Specific Hunting and Sport Fishing Regulations*—Interior FWS (rules and regulations)
  - *Migratory Bird Subsistence Harvest in Alaska*—Interior FWS (rules and regulations)
  - *Migratory Bird Hunting*—Interior FWS (rules and regulations)

#### 5. Actions Related to Effective Dates of Previously Published Rules

- *Delay of Effective Date*—DoD/GSA/NASA (rules and regulations); FDA (rules and regulations)
- *Confirmation of Effective Date*—FAA (rules and regulations)
- Notice of Confirmation of Effective Date
  - *Confirmation of Effective Date of Direct Final Rule*—DOT PHMSA (rules and regulations)
  - *Direct Final Rule; Confirmation of Effective Date*—FDA (rules and regulations)
- *Notice of Stay of Action*—FDA (rules and regulations); EPA (rules and regulations)
- *Extension of Compliance Date*—FDA (notices), TSA (rules and regulations)

#### **D.** Post-Rulemaking Documents

#### 1. Special Permissions or Limited Modifications of Regulatory Requirements

- *Temporary Permit*—FDA (notice; also pre-rulemaking, as it is used to evaluate potential need for changes to existing regulations; granted to class via individual)
- *Notice of Variance*—Energy (notices)
- *Petition for Modification*—MSHA (notices)

• *Equivalency Determination*—Coast Guard (notices)

#### 2. Solicitation of Public Input at the Enforcement Stage

- *Request for Data, Information, and Views*—FDA (notices)
- *Stakeholder Meeting*—OSHA (notices)
- *Town Hall Meeting*—CMS (notices)

#### 3. Notice of Challenge to Rule

• Notice of Appeal

#### 4. Guidance

- *Request for Information*—FDA (notices; combined with a Notice of Availability of draft guidance)
- Proposed Guidance with Request for Comment—Treasury OTS (notices)
- Proposed Statement of Policy—FEC (proposed rules)
- Proposed Guidelines—NHTSA (notices)
- *Notice of Proposed Interpretation*—DHS (notices)
- *Proposed Interpretative Statement*—CFTC (proposed rules)
- *Notice of Availability*—DOT FTA (notices)
- *Proposed Generic Communication*—NRC (notice; *Federal Register* action identified as "Notice of Opportunity for Public Comment")

Agencies similarly use a variety of documents to issue final guidance:

- *Final Statement of Policy*—FDIC (notices)
- *Final Policy Statement*—NRC (rules and regulations)
- *Notice Policy Statement*—FAA (notices)
- *GDL Guidance*—FDA (not published in *Federal Register*)
- Technical Support Document—EPA
- *Final Supervisory Guidance*—Treasury FDIC (notices)
- *Industry Guidance*—FCC (notices; identified as "policy statement")
- *Notice of Final Interpretations*—Education (notices)
- Notice of Significant Guidance—USDA (notices)
- Notice of Republication of Policy Guidance With Request for Comment—DHS (notices)
- *Policy Guidance*—State (notices)
- *Notice of Policy Guidance*—DOT (rules and regulations)
- Management Directives—DHS (not published in Federal Register); NRC

# APPENDIX B



June 2019

# FEDERAL RULEMAKING

Selected Agencies Should Clearly Communicate Practices Associated with Identity Information in the Public Comment Process



Highlights of GAO-19-483, a report to congressional requesters

#### Why GAO Did This Study

Federal agencies publish on average 3,700 proposed rules yearly and are generally required to provide interested persons (commenters) an opportunity to comment on these rules. In recent vears, some high-profile rulemakings have received extremely large numbers of comments, raising questions about how agencies manage the identity information associated with comments. While the APA does not require the disclosure of identifying information from a commenter, agencies may choose to collect this information. This report examines (1) the identity information collected by Regulations.gov and agency-specific comment websites; (2) the guidance agencies have related to the identity of commenters; (3) how selected agencies treat identity information; and (4) the extent to which selected agencies clearly communicate their practices associated with identity information.

GAO selected a nongeneralizable sample of 10 federal agencies on the basis of large comment volume. GAO surveyed 52 program offices within these agencies about their comment process; and reviewed comment websites, agency guidance, and posted comment data. GAO also interviewed relevant agency officials.

#### What GAO Recommends

GAO is making a total of eight recommendations to the selected agencies to more clearly communicate to the public their policies for posting comments and associated identity information to Regulations gov and agency-specific comment websites. The selected agencies generally agreed with the recommendations.

View GAO-19-483. For more information, contact Seto J. Bagdoyan at (202) 512-6722 or bagdoyans@gao.gov.

### FEDERAL RULEMAKING

#### Selected Agencies Should Clearly Communicate Practices Associated with Identity Information in the Public Comment Process

#### What GAO Found

The Administrative Procedure Act (APA) governs the process by which many federal agencies develop and issue regulations, which includes the public comment process (see figure below).



Source: GAO. | GAO-19-483

Regulations.gov and agency-specific comment websites collect some identity information—such as name, email, or address—from commenters who choose to provide it during the public comment process. The APA does not require commenters to disclose identity information when submitting comments. In addition, agencies have no obligation under the APA to verify the identity of such parties during the rulemaking process.

GAO found that seven of 10 selected agencies have some internal guidance associated with the identity of commenters, but the substance varies, reflecting the differences among the agencies. The guidance most frequently relates to the comment intake or response to comment phases of the public comment process.

With the discretion afforded by the APA, selected agencies' treatment of commenters' identity information varies, particularly when posting duplicate comments (identical or near-identical comment text but varied identity information). Generally, officials told GAO that their agencies (1) post all comments within the comment system; or (2) maintain some comments outside of the system, such as in email file archives. For instance, one agency posts a single example of duplicate comments and indicates the total number of comments received. However, within these broad categories, posting practices vary considerably-even within the same agency-and identity information is inconsistently presented on public websites.

Selected agencies do not clearly communicate their practices for how comments and identity information are posted. GAO's key practices for transparently reporting government data state that federal government websites should disclose data sources and limitations to help public users make informed decisions about how to use the data. As a result, public users of the comment websites could reach inaccurate conclusions about who submitted a particular comment, or how many individuals commented on an issue.

# Contents

Letter		1
	Background Selected Agencies Collect Some Information from Commenters	7
	and Accept Anonymous Comments through Regulations.gov and Agency-Specific Websites	16
	Most Selected Agencies Have Some Internal Guidance Related to Commenter Identity	23
	Selected Agencies' Treatment of Identity Information Collected during the Public Comment Process Varies Selected Agencies' Practices Associated with Posting Identity Information Are Not Clearly Communicated to Public Users	30
	of Comment Websites	40
	Conclusions Recommendations for Executive Action	51 52
	Agency Comments and Our Evaluation	53
Appendix I	Survey of Program Offices with Regulatory Responsibilities	
	within Selected Agencies	57
Appendix II	Regulations.gov Comment Form Example	64
Appendix III	Electronic Comment Filing System Comment Forms	65
Appendix IV	Securities and Exchange Commission	
	Comment Form Example	67
Appendix V	Agency Comments from the Bureau of	
	Consumer Financial Protection	68
Appendix VI	Agency Comments from the Environmental Protection Agency	70

Appendix VII	Agency Comments from the Department of	
	Health and Human Services	73
Appendix VIII	Agency Comments from the Department of the Interior	75
Appendix IX	Agency Comments from the Department of Labor	77
Appendix X	Agency Comments from the Securities and Exchange Commission	80
Appendix XI	GAO Contact and Staff Acknowledgments	81
Tables		
	Table 1: Acceptable Comment Submission Types for Selected	
	Participating Agencies	13
	Table 2: Required and Optional Comment Form Fields on Regulations.gov by Agency as of December 2018	17
	Table 3: Presence of Internal Agency Identity-Related Guidance Associated with the Public Comment Process	23
	Table 4: Presence of Agency Identity-Related Guidance Associated with Comment Intake	25
	Table 5: Presence of Agency Identity-Related Guidance Associated with Comment Analysis	28
	Table 6: Presence of Agency Identity-Related Guidance	-
	Associated with Responding to Comments Table 7: Program Offices within Selected Agencies with	29
	Regulatory Responsibilities	57

#### Figures

Figure 1: The Rulemaking Process under the Administrative	
Procedure Act	8
Figure 2: The Federal Communications Commission's Public	-
Comment Submission Process as of April 2019	15
Figure 3: Example of Anonymous Comment Submitted to the	
Electronic Comment Filing System	20
Figure 4: Example of How the Centers for Medicare & Medicaid	
Services Posts Duplicate Comments	32
Figure 5: Example of How the Wage and Hour Division Posts	
Duplicate Comments	33
Figure 6: Example of How the Consumer Financial Protection	
Bureau Posts Duplicate Comments	35
Figure 7: Example of How the Securities and Exchange	
Commission Posts Duplicate Comments	36
Figure 8: The Importance of Individual Commenter Identity	
and Organizational Affiliation to Comment Analysis	38
Figure 9: Example of Additional Disclosures about Comment	
Review and Posting Provided to Commenters on a	
Comment Form	44
Figure 10: Example Food and Drug Administration Explanation	
for a Comment Withheld from Public Posting	47
Figure 11: Regulations.gov Comment Form Example	64
Figure 12: Federal Communications Commission Electronic	
Comment Filing System's Standard Filing Comment	
Form	65
Figure 13: Federal Communications Commission Electronic	
Comment Filing System's Express Filing Web Form	66
Figure 14: Securities and Exchange Commission Comment	
Form Example	67

#### Abbreviations

APA API	Administrative Procedure Act application programming interface
BLM	Bureau of Land Management
CFPB	Consumer Financial Protection Bureau
CIO	Chief Information Officer
CMS	Centers for Medicare & Medicaid Services
CSV	comma separated values
DOL	Department of Labor
EBSA	Employee Benefits Security Administration
ECFS	Electronic Comment Filing System
EPA	Environmental Protection Agency
FCC	Federal Communications Commission
FDA	Food and Drug Administration
FDMS	Federal Docket Management System
FWS	Fish and Wildlife Service
IP	Internet Protocol
NPRM	Notice of Proposed Rulemaking
PDF	Portable Document Format
PMO	Program Management Office
SEC	Securities and Exchange Commission
URL	Uniform Resource Locator
WAF	web application firewall
WHD	Wage and Hour Division

This is a work of the U.S. government and is not subject to copyright protection in the United States. The published product may be reproduced and distributed in its entirety without further permission from GAO. However, because this work may contain copyrighted images or other material, permission from the copyright holder may be necessary if you wish to reproduce this material separately.

U.S. GOVERNMENT ACCOUNTABILITY OFFICE

441 G St. N.W. Washington, DC 20548

June 26, 2019

**Congressional Requesters** 

Regulations are the means by which federal agencies establish legally binding requirements and are rooted in agencies' statutory authority. Typically, regulations require regulated parties to take specified actions or prohibit them from taking certain actions. Agencies use regulations to carry out statutory directives to achieve public policy goals such as protecting the health and safety of the public and the environment and facilitating the effective functioning of financial markets. The Administrative Procedure Act (APA) governs the process by which many federal agencies develop and issue regulations.<sup>1</sup> The APA establishes procedures and broadly applicable federal requirements for informal rulemaking, also known as notice-and-comment rulemaking.<sup>2</sup> Federal agencies publish an average of 3,700 proposed rules, or Notices of Proposed Rulemaking (NPRM), each year as part of informal rulemaking pursuant to the APA.

Among other things, the APA generally requires agencies to publish an NPRM in the *Federal Register* and provide interested persons (commenters) an opportunity to comment on the proposed rule.<sup>3</sup> Agencies must give consideration to any significant comments submitted during the comment period when drafting the final rule. This process provides, among other things, the public an opportunity to present information to agencies on the potential effects of a rule, or to suggest alternatives. To fulfill the notice-and-comment process requirements of

<sup>3</sup>The *Federal Register* is the daily journal of the federal government, and is published every business day by the National Archives and Records Administration. The *Federal Register* contains federal agency regulations, proposed rules and notices of interest to the public, and executive orders, among other things.

<sup>&</sup>lt;sup>1</sup>5 U.S.C. §§ 551-59, 701-06, 1305, 3105, 3344, 5372, 7521. The APA was originally enacted into law in 1946, Pub. L. No. 79-404, 60 Stat. 237 (1946). In addition to the requirements under the APA, an agency may also need to comply with requirements related to rulemaking imposed by other statutes.

<sup>&</sup>lt;sup>2</sup>The APA describes two types of rulemaking, formal and informal. Formal rulemaking includes a trial-type "on-the-record" proceeding, when rules are required by statute to be made on the record after opportunity for an agency hearing. In such cases, requirements under 5 U.S.C. §§ 556–557 apply. Most federal agencies use the informal rulemaking procedures outlined in 5 U.S.C. § 553, which include notice-and-comment rulemaking. The rulemaking process described in this report is informal rulemaking.

the APA, agencies may rely on Regulations.gov or their own comment websites to receive public input on proposed rules.<sup>4</sup> During the course of the notice-and-comment process, agencies may choose to collect information associated with the identity of the commenters, such as name, email, or address (identity information).<sup>5</sup>

In recent years, some high-profile rulemakings have received extremely large numbers of comments. For example, during the public comment period for an Environmental Protection Agency (EPA) 2014 rulemaking on greenhouse gas emissions, the agency reported that it received more than 4 million total comments.<sup>6</sup> Similarly, during the public comment period for the Federal Communications Commission's (FCC) 2017 Restoring Internet Freedom NPRM, FCC received more than 22 million comments through its public comment website.<sup>7</sup> Subsequently, media and others reported that some of the comments submitted to FCC were suspected to have been submitted using false identity information.<sup>8</sup>

You asked us to review issues related to identity information associated with public comments on proposed rulemakings. This report examines (1) the identity information selected agencies collect through Regulations.gov and agency-specific comment websites, (2) the internal guidance selected agencies have related to the identity of commenters, (3) how

<sup>4</sup>Regulations.gov is an interactive public website providing the general public with the opportunity to access federal regulatory information and submit comments on regulatory and nonregulatory documents published in the *Federal Register*.

<sup>5</sup>As discussed later in this report, other information may be collected by the comment websites that can be used to identify the source of comments, such as Internet Protocol (IP) addresses.

<sup>6</sup>Carbon Pollution Emission Guidelines for Existing Stationary Sources: Electric Utility Generating Units (79 Fed. Reg. 34,830 (June 18, 2014) and 80 Fed. Reg. 64,662 (Oct. 23, 2015)).

<sup>7</sup>Restoring Internet Freedom (82 Fed. Reg. 25,568 (June 2, 2017) and (83 Fed. Reg. 7,852 (Feb, 22, 2018)).

<sup>8</sup>Comments using false identity information include any comments submitted with identity information that does not accurately represent the individual submitting the comment in question. This could include anonymized names, such as "John Doe," fictitious character names, such as "Mickey Mouse," or improper use of identity information associated with a real person. As an example of the interest in such comments, the Office of the New York State Attorney General has established a specific website that allows members of the public to search FCC's comment website for comments that may have misused identity information and, if any such instances are identified, to file a consumer submission to the Attorney General's office.

selected agencies treat identity information collected during the public comment process, and (4) the extent to which selected agencies clearly communicate their practices associated with posting identity information collected during the public comment process.

To address these objectives, we selected 10 agencies (selected agencies) as case studies that received a high volume of public comments during the course of rulemaking proceedings from January 1, 2013, through December 31, 2017, including eight agencies that use Regulations.gov as their agency's comment website ("participating agencies") and two agencies that operate agency-specific comment websites ("nonparticipating agencies"). We identified agencies based on the lists of participating and nonparticipating agencies provided on Regulations.gov. Six of the selected agencies are component agencies within a larger department, as indicated below. The selected agencies are as follows:

#### **Participating Agencies**

- Bureau of Land Management (BLM), Department of the Interior;
- Centers for Medicare & Medicaid Services (CMS), Department of Health and Human Services;
- Consumer Financial Protection Bureau (CFPB);
- Employee Benefits Security Administration (EBSA), Department of Labor (DOL);
- Environmental Protection Agency (EPA);
- Fish and Wildlife Service (FWS), Department of the Interior;
- Food and Drug Administration (FDA), Department of Health and Human Services and;
- Wage and Hour Division (WHD), DOL.

#### **Nonparticipating Agencies**

- Federal Communications Commission (FCC); and
- Securities and Exchange Commission (SEC).

All 10 agencies were selected based on the total number of rulemaking comments that Regulations.gov and other agency-specific comment websites reported they received from January 1, 2013, through December
31, 2017. We selected this period to include comments submitted to rulemakings across two presidential administrations and five complete calendar years. At the time our review began, 2017 was the most recent complete calendar year. The selected agencies represent a nongeneralizable sample, and findings from this report cannot be generalized to all agencies that receive public comments as part of their rulemaking proceedings. However, as reported by Regulations.gov, the comments submitted to the eight participating agencies we selected represent more than 90 percent of all comments submitted to all participating agencies during the 5-year period.<sup>9</sup>

To select participating agencies, we obtained publicly available data from Regulations.gov for all agencies that had rulemaking dockets—the repository of documents related to a particular rulemaking—where comments were submitted from January 1, 2013, through December 31, 2017.<sup>10</sup> On the basis of the comment numbers reported by the website, we selected the eight participating agencies with more than 500,000 comments submitted to dockets that accepted comments during this time. As of March 2018, there were 128 nonparticipating agencies, most of which issued less than one NPRM per year during the 5-year period. To identify nonparticipating agencies that received a high volume of comments, we obtained a list of rules submitted to GAO for review under the Congressional Review Act from January 1, 2007, through December

<sup>10</sup>Dockets that accepted comments from January 1, 2013, through December 31, 2017, may have also received comments outside of this date range. These comments are included in the total comment count used to select participating agencies.

<sup>&</sup>lt;sup>9</sup>We made our selection of participating agencies to include agencies that received a high volume of comments based on the total number of comments as reported by Regulations.gov. We determined that the data from Regulations.gov are sufficiently reliable for the purposes of this report, to include providing us with a relative comparison of comment volume between participating agencies for the purposes of case study selection. However, in working with these data, we found that, in some cases, the total numbers as reported by Regulations.gov do not accurately reflect the total number of comments submitted to an agency. Therefore, we are not including these total numbers in this report.

31, 2017.<sup>11</sup> We identified four agencies with more than 10 rules submitted during the period (at least one rule per year). We then contacted these agencies to determine how many total comments were submitted to the agencies from January 1, 2013, through December 31, 2017, on all rulemakings. Two of the four agencies were unable to provide us with the total number of comments received over the 5-year period; accordingly, we selected the two that provided us with comment numbers, FCC and SEC. Both FCC and SEC received a number of comments comparable to the selected participating agencies. Within the 10 selected agencies, we identified 52 program offices with regulatory responsibilities and sent them survey questionnaires related to the public comment process in October 2018. All 52 program offices responded to the questionnaire, but the responses cannot be generalized to program offices outside of the selected agencies. For additional detail about the program offices we identified and survey development and administration, see appendix I.

To determine what identity information the selected agencies collect during the public comment process, we reviewed the data fields agencies require to be submitted with public comments and the optional data fields available to commenters on Regulations.gov and the agency-specific comment systems. We reviewed relevant system documentation for Regulations.gov and the agency-specific comment systems, such as user guides, system architecture documentation, and system logs. We also reviewed documentation associated with system modernization or reengineering efforts. In addition, we interviewed relevant information technology officials from the eRulemaking Program Management Office (PMO), FCC, and SEC and surveyed program offices about the information that is collected from public users of the comment systems as well as agency practices associated with anonymous comments.<sup>12</sup> Public

<sup>&</sup>lt;sup>11</sup>Congressional Review Act, Pub. L. No. 104-121, title II, subtitle E, § 251, 110 Stat. 847, 868 (Mar. 29, 1996), codified at 5 U.S.C. §§ 801–808. The statute requires all federal agencies to submit a report on each new "rule" to both houses of Congress and to the Comptroller General before it can take effect. 5 U.S.C. § 801(a)(1)(A). For purposes of the Congressional Review Act, a "rule" is defined under § 804(3). The agency must submit to the Comptroller General a complete copy of the cost-benefit analysis of the rule, if any, and information concerning the agency's actions relevant to specific procedural rulemaking requirements set forth in various statutes and executive orders governing the regulatory process. Id. § 801(a)(1)(B).

<sup>&</sup>lt;sup>12</sup>The eRulemaking PMO leads the eRulemaking Program and is responsible for the development and implementation of Regulations.gov, the public-facing comment website, and the Federal Docket Management System (FDMS), which is the agency-facing side of the comment system used by participating agencies.

users are members of the public interested in participating in the rulemaking process via Regulations.gov or agency-specific websites. They may or may not submit a comment.

To determine what internal guidance selected federal agencies have related to the identity of commenters in the federal rulemaking process, we first determined whether each of the selected agencies had any documented policies, procedures, or guidance associated with each phase of the comment process generally. For those agencies that did, we reviewed these documents to determine whether they explicitly included requirements associated with identity information. We also included questions about guidance in our survey of program offices. On the basis of the responses, we followed up directly with program offices to obtain additional informal guidance that is used at the program office level, rather than agency-wide.

To determine how selected agencies treat identity information associated with public comments, our survey of all 52 program offices with regulatory responsibilities included questions about their practices associated with comment intake (including identifying duplicate comments and posting comments to the public website), comment analysis (including reviewing comments and considering their content), and response to comments. We obtained comment data for all rulemakings within the 10 selected agencies that accepted comments from January 1, 2013, through December 31, 2017, and reviewed them to determine the ways in which agencies treat the identity information submitted with comments. To assess the reliability of these data, we reviewed related documentation, interviewed knowledgeable agency officials, and traced selections to the source documents. We determined these data to be reliable for the purposes of selecting case study agencies and identifying comments that could help us understand how the selected agencies publicly post comments. We also interviewed relevant officials at the selected agencies, as well as officials from the eRulemaking PMO, to better understand the data; Regulations gov and FDMS; and agency-specific comment systems.

To determine the extent to which the selected agencies clearly communicate their practices associated with posting identity information collected during the public comment process, we reviewed Regulations.gov, agency-specific comment websites, and the selected agencies' websites for any information provided to public users. We then

	compared this information to key practices for transparently reporting open government data. <sup>13</sup> We also interviewed relevant officials from the eRulemaking PMO and the selected agencies about how they communicate with public users.
	We conducted this performance audit from February 2018 through June 2019 in accordance with generally accepted government auditing standards. Those standards require that we plan and perform the audit to obtain sufficient, appropriate evidence to provide a reasonable basis for our findings and conclusions based on our audit objectives. We believe that the evidence obtained provides a reasonable basis for our findings and conclusions based on our audit objectives.
Background	
The Rulemaking Process under the APA	Under the APA, agencies engage in three basic phases of the rulemaking process: they initiate rulemaking actions, develop proposed rulemaking actions, and develop final rulemaking actions. Built into agencies' rulemaking processes are opportunities for internal and external deliberations, reviews, and public comments. Figure 1 provides an overview of the rulemaking process.

<sup>&</sup>lt;sup>13</sup>GAO, Open Data: Treasury Could Better Align USAspending.gov with Key Practices and Search Requirements, GAO-19-72 (Washington, D.C.: Dec. 13, 2018).



Figure 1: The Rulemaking Process under the Administrative Procedure Act

Source: GAO. | GAO-19-483

Note: Not all agencies are required to participate in all steps for all rulemaking proceedings.

The public comment portion of the rulemaking process generally comprises three phases:

1. **Comment Intake:** During this phase, agencies administratively process comments. This may include identifying duplicate comments (those with identical or near-identical comment text, but unique identity information), posting comments to the agency's public website, and distributing comments to agency subject-matter experts within responsible program offices for analysis.

	2. <b>Comment Analysis:</b> During this phase, subject-matter experts analyze and consider submitted comments. This may include the use of categorization tools within FDMS or outside software systems. <sup>14</sup>
	3. <b>Comment Response:</b> During this phase, agencies prepare publicly available responses to the comments in accordance with any applicable requirements. Agencies are required to provide some response to the comments in the final rule, but in some cases, an agency may also prepare a separate report to respond to the comments.
Legal Requirements for Public Comments	As illustrated in figure 1 above, the public has the opportunity to provide input during the development of agencies' rules. Among other things, the APA generally requires agencies to
	• publish an NPRM in the Federal Register,
	<ul> <li>allow any interested party an opportunity to comment on the rulemaking process by providing "written data, views, or arguments";</li> </ul>
	<ul> <li>issue a final rule accompanied by a statement of its basis and purpose; and</li> </ul>

<sup>&</sup>lt;sup>14</sup>Categorization tools can include manual coding using common software such as Excel, or automated software that can group comments by issue area or keywords.

publish the final rule at least 30 days before it becomes effective.<sup>15</sup>

The APA requires agencies to allow any interested party to comment on NPRMs. The APA does not require the disclosure of identifying information from an interested party that submits a comment. Agencies therefore have no obligation under the APA to verify the identity of such parties during the rulemaking process. Instead, the APA and courts require agencies to consider relevant and substantive comments, and agencies must explain their general response to them in a concise overall statement of basis and purpose, which in practice forms part of the preamble of the final rule.<sup>16</sup> Courts have explained that significant comments are comments that raise relevant points and, if true or if adopted, would require a change in the proposed rule.<sup>17</sup> However, courts have held that agencies are not required to respond to every comment individually.<sup>18</sup> Agencies routinely offer a single response to multiple identical or similar comments. As explained by Regulations.gov's "Tips for Submitting Effective Comments," "the comment process is not a vote," and "agencies make determinations for a proposed action based on sound reasoning and scientific evidence rather than a majority of votes. A

<sup>16</sup>5 U.S.C. § 553(c). Pursuant to 1 C.F.R. § 18.12(c), an agency may include in the preamble, as applicable: a discussion of the background and major issues involved; any significant differences between the final rule and the proposed rule; a response to substantive public comments received; and any other information the agency considers appropriate.

<sup>17</sup>Safari Aviation Inc. v. Garvey, 300 F.3d 1144, 1151 (9th Cir. 2002); Am. Min. Congress v. EPA, 907 F.2d 1179, 1188 (D.C. Cir. 1990).

<sup>&</sup>lt;sup>15</sup>The APA allows agencies to issue final rules without the use of an NPRM under various exceptions, such as those dealing with military or foreign affairs and agency management or personnel. 5 U.S.C. § 553(a). APA requirements to publish a proposed rule generally also do not apply when an agency finds, for "good cause," that those procedures are "impracticable, unnecessary, or contrary to the public interest." 5 U.S.C. § 553(b). In such cases, agencies may solicit comments through the *Federal Register* when publishing a final rule without an NPRM, but the public does not have an opportunity to comment before the rule's issuance, nor is the agency obligated to respond to comments it has received. In 2012, we reported that agencies did not publish an NPRM, enabling the public to comment on a proposed rule, for about 35 percent of major rules (generally those rules with an annual impact on the economy of \$100 million or more) and about 44 percent of nonmajor rules published during 2003 through 2010. See GAO, *Federal Rulemaking: Agencies Could Take Additional Steps to Respond to Public Comments*, GAO-13-21 (Washington, D.C.: Dec. 20, 2012).

<sup>&</sup>lt;sup>18</sup>Am. Min. Congress v. EPA, 907 F.2d 1179, 1187 (D.C. Cir. 1990) (citing Thompson v. Clark, 741 F.2d 401, 408 (D.C. Cir. 1984)). See also, Action on Smoking and Health v. C.A.B., 699 F.2d 1209, 1216 (D.C. Cir. 1989).

single, well-supported comment may carry more weight than a thousand form letters."

	The APA includes provisions on the scope of judicial review that establishes the bases under which a court shall find an agency's action unlawful. <sup>19</sup> Among these APA bases are when the court finds that agency action, findings, and conclusions were "arbitrary, capricious, an abuse of discretion, or otherwise not in accordance with law" and "without observance of procedure required by law. <sup>20</sup> How an agency managed and considered public comments may be relevant during judicial review. For example, one basis for a court's reversal of an agency action has been that, upon review of the statement of basis and purposes, the court concludes the agency failed to consider or respond to relevant and significant comments. <sup>21</sup> Conversely, courts have upheld agency rules when the courts have found the statement of basis and purposes demonstrate the agency considered the commenter's arguments. <sup>22</sup>
The E-Government Act of 2002	The E-Government Act of 2002 requires agencies, to the extent practical, to accept comments "by electronic means" and to make available online the public comments and other materials included in the official rulemaking docket. <sup>23</sup> Executive Order 13563 further states that regulations should be based, to the extent feasible, on the open exchange of information and perspectives. To promote this open exchange, to the extent feasible and permitted by law, most agencies are required to provide the public with a meaningful opportunity to participate in the regulatory process through the internet, to include timely online
	<sup>19</sup> 5 U.S.C. §§ 701–706. The APA provisions govern judicial review unless the enabling statute for a particular regulatory program specifies a different standard.
	<sup>20</sup> 5 U.S.C. § 706(2)(A), (D).
	<sup>21</sup> Indep. U.S. Tanker Owners Comm. v. Lewis, 690 F.2d 908, 919 (D.C. Cir. 1982); La. Fed. Land Bank Ass'n, FLCA v. Farm Credit Admin., 336 F.3d 1075, 1080 (D.C. Cir. 2003).
	<sup>22</sup> City of Waukesha v. EPA, 320 F.3d 228, 258 (D.C. Cir. 2003).
	<sup>23</sup> Pub. L. No. 107-347, § 206, 116 Stat 2899, 2915–2916 (2002), codified at 44 U.S.C. § 3501 note.

access to the rulemaking docket in an open format that can be easily searched and downloaded.  $^{\rm 24}$ 

Most agencies meet these responsibilities through Regulations.gov, a rulemaking website where users can find rulemaking materials and submit their comments, but all agencies are not required to use that platform. In October 2002, the eRulemaking Program was established as a cross-agency E-Government initiative and is currently based within EPA. The eRulemaking PMO leads the eRulemaking Program and is responsible for developing and implementing Regulations.gov, the public-facing comment website, and FDMS, which is the agency-facing side of the comment system used by participating agencies.<sup>25</sup>

As of March 2018, Regulations.gov identified 180 participating and 128 nonparticipating agencies. These agencies may be components of larger departmental agencies. Some nonparticipating agencies, including FCC and SEC, have their own agency-specific websites for receiving public comments. The comment systems within the scope of this report are as follows:

• FDMS and Regulations.gov: FDMS is federal government-wide document management system structured by dockets (or file folders) that offer an adaptable solution to service a wide range of regulatory activities routinely performed by federal agencies.<sup>26</sup> The public-facing website of FDMS is Regulations.gov, which is an interactive website that allows the public to make comments on regulatory documents, review comments submitted by others, and access federal regulatory information. Regulations.gov allows commenters to submit comments to rulemakings by entering information directly in an electronic form on the Regulations.gov website. This form also allows commenters to attach files as part of their comment submission, and can be

<sup>24</sup>Exec. Order No. 13,563, § 2(b), 76 Fed. Reg. 3,821 (Jan. 18, 2011). However, this Executive Order does not apply to independent regulatory agencies such as the FCC, SEC, and CFPB.

<sup>25</sup>According to the Director of the PMO, as of January 2019, efforts are under way to move the PMO from EPA to the General Services Administration, though the overall governance structure will remain unchanged. The Director of the PMO expects the transition to be completed by October 1, 2019.

<sup>26</sup>Many of the eRulemaking partner agencies rely on FDMS and Regulations.gov to support a number of activities beyond rulemaking for public viewing or to solicit public comment such as publication of guidance documents, agency directives, policy interpretations, and Paperwork Reduction Act notices.

customized by each participating agency. Appendix II provides an example of one comment form from Regulations.gov. Additionally, all participating agencies allow comments to be submitted by mail or hand delivery. At their discretion, some participating agencies also allow comments to be submitted via email. See table 1.

# Table 1: Acceptable Comment Submission Types for Selected Participating Agencies

Agency	Regulations.gov	Email	Mail or hand delivery
Bureau of Land Management (Department of the Interior)	Yes	No	Yes
Centers for Medicare & Medicaid Services (Department of Health and Human Services)	Yes	No	Yes
Consumer Financial Protection Bureau	Yes	Yes	Yes
Employee Benefits and Security Administration (Department of Labor)	Yes	Yes	Yes
Environmental Protection Agency	Yes	Yes	Yes
Fish and Wildlife Service (Department of the Interior)	Yes	No	Yes
Food and Drug Administration (Department of Health and Human Services)	Yes	No	Yes
Wage and Hour Division (Department of Labor)	Yes	No	Yes

Source: GAO. | GAO-19-483

 FCC's Electronic Comment Filing System (ECFS): ECFS is a webbased application that allows anyone with access to the internet to submit comments to FCC rulemaking proceedings. ECFS allows commenters to submit comments to rulemakings through two main avenues: brief text comments submitted as Express filings, and longform comments submitted as Standard filings.<sup>27</sup> Both types of filings

<sup>&</sup>lt;sup>27</sup>FCC's ECFS also accepts comments that are not part of the rulemaking process. In addition to Express and Standard filings, users can submit Non-Docketed filings in response to certain FCC proceedings that have not been assigned a docket number or rulemaking number.

can be submitted through an ECFS comment form, which requires commenters to enter information directly into an electronic form on the ECFS website. See appendix III for examples of the comment forms used by ECFS. Additionally, interested parties with the appropriate technical capabilities can submit either type of filing directly to ECFS via a direct application programming interface (API) or through a public API that is registered with the website Data.gov.<sup>28</sup> Filing comments through an API allows interested parties the ability to file a large number of comments without having to submit multiple individual comment forms. Finally, to accommodate a large volume of comment submissions for the 2015 Open Internet rulemaking, FCC allowed interested parties to submit Express comment filings in bulk through formatted CSV files that were submitted via a dedicated email address and then uploaded into ECFS.<sup>29</sup> Similarly, for the 2017 Restoring Internet Freedom rulemaking, FCC allowed commenters to submit Express comment filings in bulk through a dedicated filesharing website, and the comments were then uploaded into ECFS. With the exception of these two rulemakings, FCC does not allow comments to be submitted electronically outside of ECFS. Figure 2 shows how ECFS facilitates public commenting by using the processes discussed above.

<sup>&</sup>lt;sup>28</sup>An API sets up machine-to-machine communication and allows users to connect directly with a website to provide or access data. Data.gov is a data catalog for a variety of U.S. government data sets, managed and hosted by the General Services Administration's Technology Transformation Service.

<sup>&</sup>lt;sup>29</sup>CSV, or comma separated values, is a simple format for representing a rectangular array (matrix) of numeric and textual values. It is a delimited data format that has fields/columns separated by a comma character and records/rows/lines separated by characters indicating a line break.



## Figure 2: The Federal Communications Commission's Public Comment Submission Process as of April 2019

Source: GAO analysis of FCC information. | GAO-19-483

<sup>a</sup>This option was only available for two proceedings: the 2015 Open Internet rulemaking and the 2017 Restoring Internet Freedom rulemaking. CSV, or comma separated values, is a simple format for representing a rectangular array (matrix) of numeric and textual values.

 SEC's Comment Letter Log: When SEC requests public comments on SEC rule proposals, the public can submit comments to rulemakings through an online form, which requires commenters to enter information in an electronic form on SEC's website. This form also allows commenters to attach files as part of their submission. When commenters submit a comment, it is sent to SEC staff as an email. SEC also allows comments to be submitted via email and mail. After review, staff upload the comment and any associated data into the Comment Letter Log, which is the internal database that SEC staff use to manage the public comment process, and post the comment to the public website. See appendix IV for an example of a comment form on SEC's website.

Selected Agencies Collect Some Information from Commenters and Accept Anonymous Comments through Regulations.gov and Agency-Specific Websites	
Selected Agencies Collect Some Identity Information through Comment Forms	Consistent with the discretion afforded by the APA, Regulations.gov and agency-specific comment websites use required and optional fields on comment forms to collect some identity information from commenters. In addition to the text of the comment, each participating agency may choose to collect identity information from the Regulations.gov comment form by requiring commenters to fill in other fields, such as name, address, and email address before they are able to submit a comment. Participating agencies may also choose to collect additional identity information through optional fields. <sup>30</sup> For example, while EPA does not make any fields associated with identity information available to commenters, CFPB makes all fields available and requires that commenters enter something into the first name, last name, and organization name fields before a comment can be submitted. Table 2 shows the fields on Regulations.gov in which each of the participating agencies we analyzed require commenters to voluntarily enter information.
	<sup>30</sup> As of April 2019, the eRulemaking Advisory Board is considering modifications to Regulations.gov comment form requirements as part of a broader modernization effort.

Regulations.gov comment form requirements as part of a broader modernization effort. Specifically, according to PMO officials, one goal for the modernized Regulations.gov/FDMS will be to include a standardized comment form for use by all participating agencies. As of April 2019, this new form is being internally tested by the eRulemaking PMO, and also requires commenters to choose whether they are an individual, an organization, or an anonymous entity before they are able to submit the comment. However, the PMO is expected to transition its operational programs from its current home in EPA to the General Services Administration in the fall of 2019. As a result, PMO officials stated that proposed modifications from the modernization will likely not be addressed until after the PMO operational transition is complete.

### Table 2: Required and Optional Comment Form Fields on Regulations.gov by Agency as of December 2018

	Agency							
Field	Bureau of Land Management	Centers for Medicare & Medicaid Services	Consumer Financial Protection Bureau	Employee Benefits Security Administration	Environmental Protection Agency	Fish and Wildlife Service	Food and Drug Administration	Wage and Hour Division
Comment	Required	Required	Required	Required	Required	Required	Required	Required
First Name	Optional	Optional	Required	Optional	n/a	Optional	Optional	Required
Middle Name	Optional	n/a	Optional	Optional	n/a	Optional	Optional	Optional
Last Name	Optional	Optional	Required	Optional	n/a	Optional	Optional	Required
Mailing Address	Optional	n/a	Optional	Optional	n/a	Optional	n/a	Required
City	Optional	Optional	Optional	Optional	n/a	Optional	n/a	Required
State or Province	Optional	Required	Optional	Optional	n/a	Optional	n/a	Required
Zip/Postal Code	Optional	Required	Optional	Optional	n/a	Optional	Optional	Optional
Country	Optional	Required	Optional	Optional	n/a	Optional	n/a	Required
Email Address	Optional	Optional	Optional	Optional	n/a	Optional	Optional	Optional
Phone Number	Optional	n/a	Optional	Optional	n/a	Optional	n/a	Optional
Fax Number	Optional	n/a	Optional	n/a	n/a	Optional	n/a	n/a
Upload Files	Optional	Optional	Optional	Optional	Optional	Optional	Optional	Optional
Submitter's Representative	Optional	n/a	Optional	n/a	Optional	Optional	Optional	n/a
Organization Name	Optional	Required	Required	Optional	Optional	Optional	Required	Optional
Government Agency Type (i.e., Federal/ State/Local)	Optional	n/a	Optional	n/a	Optional	Optional	n/a	n/a
Government Agency	Optional	n/a	Optional	n/a	Optional	Optional	n/a	n/a
Category (e.g., Academia, Consumer Group, Individual Consumer)	n/a	Optional	n/a	n/a	n/a	Optional	Required	n/a

Source: eRulemaking Program Management Office. | GAO-19-483

Note: "n/a" indicates that a field is not available to a commenter on the agency's electronic comment forms.

	FCC requires that all commenters complete the following fields on both the Standard and Express comment forms in ECFS: (1) name, (2) postal address, and (3) the docket proceeding number to which they are submitting a comment. <sup>31</sup> The ECFS comment form also allows commenters to voluntarily provide additional information in optional fields, such as email address. Similarly, SEC's comment forms require commenters to provide (1) first and last name, (2) email address, and (3) the comment content, before a comment can be successfully submitted. The comment form also allows commenters to voluntarily provide other information in optional fields, such as their city and state.
Agencies Accept Anonymous Comments	Regardless of the fields required by the comment form, the selected agencies all accept anonymous comments in practice. Specifically, in the comment forms on Regulations.gov, ECFS, and SEC's website, a commenter can submit a comment under the name "Anonymous Anonymous," enter a single letter in each required field, or provide a fabricated address. In each of these scenarios, as long as a character or characters are entered into the required fields, the comment will be accepted. Further, because the APA does not require agencies to authenticate submitted identity information, neither Regulations.gov nor the agency-specific comment websites contain mechanisms to check the validity of identity information that commenters submit through comment forms.
	As part of the Regulations.gov modernization effort, the Office of Information and Regulatory Affairs (within the Office of Management and Budget) and the Department of Justice proposed language for a disclosure statement on every comment form that would require the commenter to acknowledge that they are not using, without lawful authority, a means of identification of another person with any comment they are submitting. Commenters would be required to acknowledge their agreement with the statement before their comment could be submitted.
	<sup>31</sup> As of April 2019, FCC is undertaking a reengineering effort to update ECFS, beginning with a discovery phase that involves, among other things, identifying system requirements that will help FCC meet APA requirements and improve the security of ECFS. According to FCC's Chief Information Officer (CIO), the discovery phase of the ECFS reengineering process began in the first quarter of fiscal year 2019 and is expected to be completed in May 2019. After the discovery phase, FCC will move to an implementation phase, which will include awarding a contract for the project, development and implementation of the new system, and going live with the new system. According to the CIO, the new system is expected to be completed by April 2020.

According to PMO officials, even with this disclosure statement, anonymous comments would still be permitted and accepted by Regulations.gov. This disclosure statement was proposed in response to allegations of comments being submitted to rulemakings on behalf of individuals without their permission. As of April 2019, this proposed language has not yet been approved by the Executive Steering Committee for Regulations.gov. However, the proposed disclosure statement would be provided on the Regulations.gov comment form, and it is unclear whether similar information would be made available to commenters submitting comments via email or mail.

In contrast to the other selected agencies, according to FCC officials, FCC rules require the submission of the commenter's name and mailing address, or the name and mailing address of an attorney of record. However, in March 2002, FCC initiated a rulemaking related to the submission of truthful statements to the commission.<sup>32</sup> Among other issues, FCC sought comment on whether rulemaking proceedings should be subject to an already existing rule that prohibited the submission of written misrepresentations or material omissions from entities that are subject to FCC regulation. In its final rule, issued in March 2003, FCC decided to continue to exempt comments to rulemakings from this rule because of the potential that such a requirement would hinder full and robust public participation in such policy-making proceedings by encouraging disputes over the truthfulness of the parties' statements.<sup>33</sup> According to FCC officials, to comply with APA requirements, the commission tries to minimize barriers that could prevent or discourage commenters from participating in the commenting process, and in practice accepts anonymous comments. See figure 3 for an example of an anonymous comment in ECFS.

<sup>&</sup>lt;sup>32</sup>67 Fed. Reg. 10,658 (Mar. 8, 2002).

<sup>&</sup>lt;sup>33</sup>68 Fed. Reg. 15,096 (Mar. 28, 2003).



### Figure 3: Example of Anonymous Comment Submitted to the Electronic Comment Filing System

Source: GAO analysis of ECFS information. | GAO-19-483

Note: We did not include the identity information associated with these comments other than the address, which is obviously fabricated.

Additionally, in our survey of program offices with rulemaking responsibilities at selected agencies, 39 of 52 offices reported that they received anonymous comments on some rulemakings for which their office has been responsible since 2013. The remaining 13 offices responded that they did not receive or were unaware of receiving anonymous comments, though most of these offices do not have high levels of rulemaking activity or receive a high volume of comments.<sup>34</sup>

<sup>&</sup>lt;sup>34</sup>Specifically, eight of the 13 program offices reported that the largest rulemaking for which they were responsible during the period received fewer than 500 comments. Further, nine of the 13 also reported that their office solicits comments on five or fewer rulemaking proceedings in a typical year.

# Regulations.gov and Agency-Specific Comment Websites Collect Some Information about Public Users' Interaction

Regulations.gov and agency-specific comment websites also collect some information about public users' interaction with their websites through application event logs and proxy server logs.<sup>35</sup> This information, which can include a public user's Internet Protocol (IP) address, browser type and operating system, and the time and date of webpage visits, is collected separately from the comment submission process as part of routine information technology management of system security and performance. The APA does not require agencies to collect or verify this type of information as part of the rulemaking process.

Regulations.gov collects some information from commenters accessing the website but it is never linked to any specific comment. In Regulations.gov, proxy server logs capture information such as the country from which a user accesses the site, the user's browser type and operating system, and the time and date of each page visit on the website. According to PMO officials, this information is provided to the eRulemaking PMO in summary statistics that are used to assess what information is of least interest to Regulations.gov visitors, determine technical design specifications of the website, and identify system performance problems. This information is collected about all public users visiting Regulations.gov, regardless of whether they submit a comment. Further, because the PMO receives this information in the form of summary statistics, it cannot be connected to any specific comment. The eRulemaking PMO also monitors IP addresses that interact with Regulations.gov via security firewalls, but, according to PMO officials, the web application firewall (WAF) logs (a type of application event log) have never been connected to specific comments, though in some cases the

<sup>&</sup>lt;sup>35</sup>Application event logs are generated by applications running on servers, end-user devices, or the web. One type of application event log is a web application firewall (WAF) log, which logs information about adverse events that triggered the firewall, including the type of threat identified by the firewall, date/time stamp, IP address, and relevant Uniform Resource Locator (URL). A proxy server log contains requests made by users and applications on a network. Proxy servers provide an application-level gateway intermediary typically between a user client web browser seeking resources from application servers. The proxy server may provide services related to, for example, implementing internet access controls such as in blocking access to websites with known security risks.

URL the blocked user was attempting to access may be captured in the log.<sup>36</sup>

FCC officials stated that the current ECFS application architecture does not facilitate FCC identifying the source IP address of the submitter of a specific comment filed in ECFS. FCC collects information about public users' interactions with ECFS through its web-based application proxy server logs, including the IP address from which a user accesses the site and the date and time of the user's interaction. However, ECFS does not obtain or store IP addresses as part of the comment data it collects when a public user ultimately submits a comment. Within the current architecture, ECFS would require officials to match date and time stamps from the proxy server log to the ECFS comment data to connect a given IP address to a specific comment.<sup>37</sup>

SEC officials stated it would be difficult to match the large number of daily hits to their general website to the much smaller number of comments submitted to their rulemaking proceedings. SEC collects information about public users' interactions with the SEC.gov website through proxy server logs, including the IP address from which a user accesses the website and the user's date, time, and URL requests. However, according to officials, a public user never directly interacts with the Comment Letter Log, and none of the information from the proxy log is included as part of the data it collects in association with comment submissions.<sup>38</sup> Despite this difficulty, SEC officials stated that linking the proxy log data from the general SEC.gov website to a specific comment in the Comment Letter Log could be done on a case-by-case basis.

<sup>37</sup>Until the discovery phase of the ECFS reengineering effort is completed in May 2019, officials could not comment on whether this issue would be addressed by the new system.

<sup>38</sup>Specifically, when a commenter submits a comment using the comment form on SEC.gov, officials receive that comment and its associated identity information as an email, which is then entered into the Comment Letter Log by SEC staff.

<sup>&</sup>lt;sup>36</sup>An IP address is a code that identifies a computer network or a particular device on a network, consisting of four numbers separated by periods. There are many ways to obscure IP addresses, such as by using a Virtual Private Network, which is a program that creates a safe and encrypted connection over a less secure network, such as the public internet. A URL is the address of a resource on the internet. According to PMO officials, the firewalls are monitored for IP address activity that may be indicative of a denial of service attack, which is a cyberattack in which the perpetrator seeks to make a machine or network resource unavailable to its intended users by temporarily or indefinitely disrupting services of a host connected to the internet.

# Most Selected Agencies Have Some Internal Guidance Related to Commenter Identity

Seven of 10 selected agencies have documented some internal guidance associated with the identity of commenters during the three phases of the public comment process, but the substance of this guidance varies, reflecting the differences among the agencies and their respective program offices.<sup>39</sup> For example, as shown in table 3, BLM has no internal guidance related to identity information, while CFPB has internal guidance related to the comment intake and response to comments phases.

# Table 3: Presence of Internal Agency Identity-Related Guidance Associated with the Public Comment Process

Agency	Comment intake	Comment analysis	Response to comments
Bureau of Land Management	No	No	No
Centers for Medicare & Medicaid Services	No	Yes	Yes
Consumer Financial Protection Bureau	Yes	No	Yes
Employee Benefits Security Administration <sup>a</sup>	Yes	Yes	Yes
Environmental Protection Agency	Yes	No	No
Federal Communications Commission	No	No	No
Fish and Wildlife Service	No	No	No
Food and Drug Administration	No	Yes	Yes
Securities and Exchange Commission	Yes	No	No
Wage and Hour Division <sup>a</sup>	Yes	Yes	Yes

Source: GAO analysis of agency data. | GAO-19-483

<sup>a</sup>The Employee Benefits Security Administration and Wage and Hour Division provided GAO with Department of Labor guidance that applies to all agencies within the department.

<sup>&</sup>lt;sup>39</sup>As used in this report, "guidance" refers to documented items such as internal standard operating procedures and training materials designed to assist agency staff in carrying out their daily responsibilities. We are not referring to formal guidance documents, defined by the Office of Management and Budget as an agency statement of general applicability and future effect, other than a regulatory action, that sets forth a policy on a statutory, regulatory, or technical issue, or an interpretation of a statutory or regulatory issue.

	For selected agencies that have guidance associated with the identity of commenters, it most frequently relates to the comment intake or response to comment phases of the public comment process. The guidance for these phases addresses activities such as managing duplicate comments (those with identical or near-identical comment text but varied identity information) or referring to commenters in a final rule. In addition, some agencies have guidance related to the use of identity information during comment analysis. Agencies are not required by the APA to develop internal guidance associated with the public comment process generally, or identity information specifically. For the three selected agencies that did not have identity-related guidance has not been developed because identity information is not used as part of their rulemaking process. For example, BLM officials stated that the only instance in which identity information would be considered is when threatening comments are referred to law-enforcement agencies.
Identity-Related Guidance for Comment Intake	According to our analysis of the internal guidance the selected agencies provided, five of the 10 agencies have documented identity-related guidance associated with the comment intake phase. (See table 4.) Identity-related guidance for the comment intake phase addresses posting comments and their associated identity information to public comment websites. The guidance generally falls into two categories: (1) the treatment of duplicate comments (those comments with identical or near-identical content, but unique identity information) and (2) the management of comments reported to have been submitted using false identity information.

# Table 4: Presence of Agency Identity-Related Guidance Associated with Comment Intake

Agency	Duplicate comments	Comments with potentially false identity information
Bureau of Land Management	No	No
Centers for Medicare & Medicaid Services	No	No
Consumer Financial Protection Bureau	Yes	Yes
Employee Benefits Security Administration <sup>a</sup>	Yes	Yes
Environmental Protection Agency	Yes	Yes
Federal Communications Commission	No	No
Fish and Wildlife Service	No	No
Food and Drug Administration	No	No
Securities and Exchange Commission	No	Yes
Wage and Hour Division <sup>a</sup>	Yes	Yes

#### Source: GAO analysis of agency data. | GAO-19-483

<sup>a</sup>The Employee Benefits Security Administration and Wage and Hour Division provided GAO with Department of Labor guidance that applies to all agencies within the department.

## **Duplicate Comments**

Four of the 10 selected agencies have documented guidance on defining and posting duplicate comments, which may also be referred to as mass mail campaigns.<sup>40</sup> However, in accordance with the discretion afforded them under the APA, agency definitions of duplicate comments and recommendations on how to manage them during comment intake vary.<sup>41</sup> Specifically, for EBSA and WHD—the selected agencies within the Department of Labor (DOL)—one comment letter with multiple signers is considered one comment, while the same comment submitted by multiple signers as separate letters is counted separately. In both cases, however, each individual signer may provide unique identity information. In contrast, EPA guidance states that mass mail submissions often include attachments containing either bundled duplicate messages or a single

<sup>&</sup>lt;sup>40</sup>For example, EPA established a document subtype of "Mass Mail Campaign" in FDMS, and refers to duplicate comments as those belonging to mass mail campaigns.

<sup>&</sup>lt;sup>41</sup>As discussed previously in this report, the APA requires that agencies consider the significant issues raised in the comments, not the total number of comments received in favor of or opposed to a particular rulemaking proceeding.

comment with multiple signatures. For EPA, each signature is counted as a duplicate comment submission.

As of February 2019, CFPB's draft guidance does not explicitly define duplicate comments, but it does note that "duplicate identical submissions" are not subject to the agency's policy of posting all comments. Instead, the official responsible for managing the docket during comment intake may remove duplicate comments from posting or decide not to post them. According to CFPB officials, this policy is only applicable to comments that contain entirely identical comment content and identity information, and does not apply to mass mailing campaigns. Similarly, when DOL agencies receive duplicate comments as part of mass mail campaigns, the agency can choose to post a representative sample of the duplicate comment to Regulations gov along with the tally of the duplicate or near-duplicate submissions, or post all comments as submitted. EPA guidance states that duplicate comments submitted as part of mass mailings are to be posted as a single primary document in Regulations gov with a tally of the total number of duplicate comments received from that campaign. However, as discussed later in this report, EPA may post all duplicate comments it receives, depending on the format in which they are submitted.

Comments with Potentially False Identity Information Five of the 10 selected agencies have documented internal guidance on how to manage posting comments that may have been submitted by someone falsely claiming to be the commenter. However, the procedures related to addressing comments with potentially false identity information also vary among agencies. For EBSA and WHD, guidance from DOL states that if a comment was submitted by someone falsely claiming to be the commenter, the identifying information is to be removed from the comment and the comment is treated as anonymous and remains posted.<sup>42</sup> In cases where an individual claims that a comment was submitted to CFPB or SEC using the individual's identity information without his or her consent, both agencies' guidance provide staff with discretion to redact, reattribute, or otherwise anonymize the comment letter in question.<sup>43</sup>

<sup>&</sup>lt;sup>42</sup>DOL guidance does not indicate how officials determine that a comment was submitted using false identity information.

<sup>&</sup>lt;sup>43</sup>CFPB's guidance is in draft form as of February 2019, and may be subject to further revision.

	According to internal guidance from CFPB, EPA, and SEC, if agency officials are able to confirm that a comment was submitted by someone falsely claiming to be the commenter, such as by the agency sending an email to the address associated with the comment, the comment may not be made available to the public. SEC officials stated that although they have discretion to remove the comment from public posting, the typical response is to encourage the individual making the claim to submit another comment correcting the record. Similarly, if a member of the public contacts EPA claiming that a comment was submitted using his or her identity information without consent and agency staff cannot confirm it, EPA guidance directs staff to ask the requester who submitted the claim to submit another comment to the docket explaining that the original comment was submitted without the individual's consent. Both comments will be included in the docket. <sup>44</sup>
Identity-Related Guidance for Comment Analysis	According to our analysis of the guidance the selected agencies provided, four of the 10 agencies have identity-related guidance for the comment analysis phase (see table 5). Identity-related guidance for the comment analysis phase includes criteria for coding comments for analysis, including by identifying the type of commenter (such as an individual or interest group).

<sup>&</sup>lt;sup>44</sup>According to FCC officials, the agency developed a similar policy for the Restoring Internet Freedom NPRM. Specifically, when FCC received a claim from a member of the public that a comment was submitted to the Restoring Internet Freedom NPRM using his or her identity information without the individual's consent, officials directed the individual to enter the complaint as a new comment, but did not remove the original comment.

Table 5: Presence of Agency Identity-Related Guidance Associated with Comment	
Analysis	

Agency	Comment analysis
Bureau of Land Management	No
Centers for Medicare & Medicaid Services	Yes
Consumer Financial Protection Bureau	No
Employee Benefits Security Administration <sup>a</sup>	Yes
Environmental Protection Agency	No
Federal Communications Commission	No
Fish and Wildlife Service	No
Food and Drug Administration	Yes
Securities and Exchange Commission	No
Wage and Hour Division <sup>a</sup>	Yes

Source: GAO analysis of agency data. | GAO-19-483

<sup>a</sup>The Employee Benefits Security Administration and Wage and Hour Division provided GAO with Department of Labor guidance that applies to all agencies within the department.

Specifically;

- CMS guidance states that, during review, comments should be separated by issue area and tables may be used to assist in the grouping of comments and preparing briefing materials. While this guidance notes that these tables may be used to group commenters based on their identity during review, when summarizing comments later in the process the guidance indicates that CMS officials should avoid identifying commenters by name or organization.
- FDA training materials address how to prepare comment summaries to help ensure the agency has properly identified all comments regarding an issue. To conduct a quality-control check on the comment review process, FDA sorts the comments by commenter and reviews the comments from a sample of key stakeholders, including interested trade associations and consumer or patient groups, to confirm that relevant issues were identified.
- For EBSA and WHD, guidance from DOL recommends attaching the "organization name" to comments within a docket to improve transparency and help the agency and public users search for organizations within Regulations.gov. In addition, DOL guidance

suggests flagging comments for additional review, including at least one flag based on identity.  $^{\rm 45}$ 

Identity-Related Guidance for Responding to Comments According to our analysis of the guidance the selected agencies provided, five of the 10 agencies have documented identity-related guidance for responding to comments. (See table 6.) Identity-related guidance for the response to comments phase includes guidance for agency officials on how, if at all, to address identity information related to comments in developing the final rule.

# Table 6: Presence of Agency Identity-Related Guidance Associated with Responding to Comments

Agency	Response to comments
Bureau of Land Management	No
Centers for Medicare & Medicaid Services	Yes
Consumer Financial Protection Bureau	Yes
Employee Benefits Security Administration <sup>a</sup>	Yes
Environmental Protection Agency	No
Federal Communications Commission	No
Fish and Wildlife Service	No
Food and Drug Administration	Yes
Securities and Exchange Commission	No
Wage and Hour Division <sup>a</sup>	Yes

Source: GAO analysis of agency data. | GAO-19-483

<sup>a</sup>The Employee Benefits Security Administration and Wage and Hour Division provided GAO with Department of Labor guidance that applies to all agencies within the department.

### Specifically;

- As discussed previously, during comment analysis, CMS guidance indicates that officials should avoid identifying commenters by name or organization when summarizing comments. These summaries may then be used as a basis for the agency's formal comment summary included in the preamble of the final rule.
- CFPB guidance states that a summary of the rulemaking process should be developed for the preamble of the final rule and include

<sup>&</sup>lt;sup>45</sup>Specifically, DOL guidance instructs officials to flag any comments submitted by members of Congress.

	<ul> <li>how many comments are received and from which type of commenter. CFPB is to describe both the commenters and comments in general terms rather than identify commenters by name or entity. For example, rather than naming a specific financial institution, CFPB may refer to "industry commenters" in the final rule.</li> <li>For EBSA and WHD, guidance from DOL states that when several commenters suggest the same approach to revising or modifying the proposed rule, the names of specific commenters can be cited as a</li> </ul>
	list in a footnote. When choosing which commenter should appear first in the list, DOL agencies are to select the commenter with the strongest or most detailed discussion on the issue. However, it is not necessary to always identify commenters by name, and, according to DOL officials, the department's general practice is not to do so. Instead, the agency may use phrases such as "several commenters," or "comments by the ABC Corporation and others." DOL agencies may also reference commenters by type rather than name, using terms including "municipal agency, state workforce agency, employer, academic representative, agency, and industry," among others.
	• FDA training materials recommend that the final rule include a very brief explanation of the number and scope of comments to the proposed rule, including who submitted them. Commenters are not identified as individuals, but rather by commenter type, such as trade associations, farms, or consumer advocacy organizations, among others.
Selected Agencies' Treatment of Identity Information Collected during the Public Comment Process Varies	Within the discretion afforded by the APA, the 10 selected agencies' treatment of identity information during the comment intake, comment analysis, and response to comments phases of the public comment process varies. Selected agencies differ in how they treat identity information during the comment intake phase, particularly in terms of how they post duplicate comments, which can lead to identity information being inconsistently presented to public users of comment systems. Selected agencies' treatment of identity information during the comment analysis phase also varies. Specifically, program offices with responsibility for analyzing comments place varied importance on identity information during the analysis phase. All agencies draft a response to comments with their final rule, but the extent to which the agencies identify commenters or commenter types in their response also varies across the selected agencies.

# Selected Agencies Vary in Their Treatment of Identity Information during the Comment Intake Phase

Within the discretion afforded by the APA and E-Government Act, selected agencies vary in how they treat identity information during the comment intake phase, which includes identifying duplicate comments and posting comments to the public website. Further, the way in which the selected agencies treat comments during the comment intake phase results in identity information being inconsistently presented on the public website. Generally, officials told us that their agencies either (1) maintain all comments within the comment system, or (2) maintain some duplicate comment records outside of the comment system, for instance, in email file archives. Specifically, officials from four selected agencies (CMS, FCC, FDA, and WHD) stated that they maintain all submitted comments in the comment system they use. Officials from the other six agencies (BLM, CFPB, EBSA, EPA, FWS, and SEC) stated that their agencies maintain some comment records associated with duplicate comments outside of the comment system.

Among the four agencies that maintain all submitted comments within their comment system, our review of comment data showed that practices for posting duplicate comments led to some identity information or comment content being inconsistently presented on the public website. For example, according to CMS officials responsible for comment intake, CMS may post all duplicate comments individually, or post duplicate comments in batches. When duplicate comments are posted in batches, the comment title will include the name of the submitting organization followed by the total number of comments. However, as discussed previously, CMS does not have any documented policies or guidance associated with the comment intake process, and we identified examples where the practices described by CMS officials differed. On one CMS docket, for instance, staff entered more than 37,000 duplicate comments individually, with the commenter's name and state identified in the comment title. However, the attached document included with each of the posted comments was an identical copy of one specific comment containing a single individual's identity information. While all the individual names appear to have been retained in the comment titles, and the count of total comments is represented, any additional identity information and any potential modifications made to each duplicate comment submitted have not been retained either online or outside of FDMS, and are not presented on the public website. (See fig. 4.)



# Figure 4: Example of How the Centers for Medicare & Medicaid Services Posts Duplicate Comments

Source: GAO analysis of Regulations.gov information. | GAO-19-483

Note: We did not include the identity information associated with these comments. Instead, each unique piece of identity information is identified by a different number or letter.

Similarly, although our analysis of WHD comments did not suggest that any comments were missing from Regulations.gov, on one WHD docket almost 18,000 duplicate comments were associated with a single comment with one individual's name identified in the comment title. While all of the comments are included within 10 separate attachments, none of the identity information included with these comments can be easily found without opening and searching all 10 attachments, most of which contain approximately 2,000 individual comments. (See fig. 5.)



### Figure 5: Example of How the Wage and Hour Division Posts Duplicate Comments

Source: GAO analysis of Regulations.gov information. | GAO-19-483

Note: We did not include the identity information associated with these comments. Instead, each unique piece of identity information is identified by a different number or letter.

Our review of comment data showed that the selected agencies that maintain some comment records outside of the comment system (six of 10) also follow practices that can inconsistently present some identity information or comment content associated with duplicate comments. For BLM and FWS, agency officials responsible for comment intake stated that all comments received through Regulations.gov are posted, but a single example may be posted when duplicate paper comments are received.<sup>46</sup> As discussed previously, neither BLM nor FWS have internal guidance or policy associated with comment intake. For CFPB, EBSA, EPA, and SEC, the agency may post a single example along with the total count of all duplicate comments, but does not necessarily post all duplicate comments online. Thus, identity information and unique comment contents for all duplicate comments may not be present on the public website. For example, on one CFPB comment, the agency posted an example of a submitted comment containing only the submitter's illegible signature.<sup>47</sup> None of the other associated identity information for the posted sample, or any of the duplicate comments, is included in the comment data. (See fig. 6.)

<sup>&</sup>lt;sup>46</sup>According to BLM officials, although this is the agency's practice, there have been some instances where a technical issue with FDMS has prevented them from publicly posting all duplicate comments submitted through Regulations.gov.

<sup>&</sup>lt;sup>47</sup>According to CFPB officials, they anticipate finalizing a new policy related to how duplicate comments are treated that would result in all comments being posted.



Figure 6: Example of How the Consumer Financial Protection Bureau Posts Duplicate Comments

Source: GAO analysis of Regulations.gov information. | GAO-19-483

Similarly, for all duplicate comments received, SEC posts a single example for each set of duplicate comments and indicates the total number of comments received. As a result, the identity information and any unique comment content beyond the first example are not present on the public website.<sup>48</sup> (See fig. 7.)



Source: GAO analysis of Securities and Exchange Commission information. | GAO-19-483

The Importance of Identity Information to Comment Analysis Varies On the basis of the results from our survey, program offices with responsibility for analyzing comments differ in the importance they place on identity information during the analysis phase. Because subject-matter experts are responsible for reviewing public comments and considering whether changes to the proposed rule should be made, program offices generally analyze comments. Officials from all but one of the 52 program

<sup>48</sup>According to SEC officials, if the unique content includes an argument distinguishing it from the other duplicate comments, it will be counted and posted separately.

offices we surveyed responded that they were responsible, in whole or in part, for analyzing public comments.<sup>49</sup>

In our survey of program offices with regulatory responsibilities in the 10 selected agencies, at least one program office in each agency reported that the identity or organizational affiliation of a commenter is at least slightly important to comment analysis. Additionally, five of the 10 selected agencies (CMS, EPA, FCC, FDA, and FWS) had at least one program office that reported that the identity or organizational affiliation of a commenter is not at all important to comment analysis. None of the 52 program offices we surveyed responded that the identity of an individual commenter is extremely important to their analysis, while only one program office responded that the commenter's organizational affiliation is extremely important to its analysis. (See fig. 8.)

<sup>&</sup>lt;sup>49</sup>Officials from one program office within EPA responded that they are not responsible for analyzing public comments, but noted this responsibility is shared between program offices and a work group.

### Figure 8: The Importance of Individual Commenter Identity and Organizational Affiliation to Comment Analysis



 
 Not at all important
 Slightly important
 Moderately important
 Very important
 Extremely important

 Source: GAO analysis of survey data.
 GAO-19-483
 GAO-1

According to officials we interviewed from eight of the 10 selected agencies, the substance of the comment is considered during analysis rather than the submitted identity information.<sup>50</sup> Officials from six of these agencies emphasized that because the agency accepts anonymous comments, identity is not relevant to their analysis of comments. However, officials from four of the eight selected agencies stated that, in certain instances, identity information may be noted. In the case of FDA, officials explained that commenters are required to indicate a category to which they belong, such as "individual consumer" or "academia." According to FDA officials, however, these categories were used to assist in writing the comment response, rather than informing the analysis. Officials from the Department of the Interior's Office of the Solicitor (responsible for part of the comment process at BLM and FWS) stated

<sup>&</sup>lt;sup>50</sup>During our interview with CFPB officials at the agency level, officials did not comment on whether identity information was considered during comment analysis.

	that the agency may make particular note of comments submitted by a law firm, as these comments can help the agency understand the position of the law firm and to prepare a defense in the event that a lawsuit is filed. Similarly, officials from EPA stated that they are familiar with many commenters and their positions on certain issues, due to prior legal interactions. In another example of how an agency may consider the identity of a commenter, officials from FWS stated that when scientific data are provided in support of a comment, subject-matter experts will verify the data and their source.
Selected Agencies Differ in How They Identify Commenters When Responding to Comments	All selected agencies draft a response to comments with their final rule, but the extent to which the agencies identify commenters in their response varies. In our survey of program offices with regulatory responsibility, officials from 51 of 52 offices stated that they are responsible in whole or in part for responding to comments. <sup>51</sup> Of those responsible, at least one program office from eight of the 10 agencies (28 of 52 offices) reported that they identified comments by commenter name, organization, or comment ID number in the response to comments for at least some rulemakings since 2013. <sup>52</sup> In the case of WHD, officials we interviewed explained that when they discuss a specific comment in the preamble to the final rule, they provide the name of the organization that submitted the comment so that anyone interested in locating the response to the comment may do so easily.
	We found that EBSA and FCC also identified commenters by individual or organizational name in their response to comments, while EPA referred to comments by their comment ID number. For example, in a rule finalized in 2018, EPA referred to comment ID numbers in the response to comments: "Two comments: EPA-R06-RCRA-2017-0556-0003 and EPA-R06-RCRA-2017-0556-0005 were submitted in favor of the issuance of the petition." EPA officials noted that there is variation within the agency in terms of how commenters are identified when the agency is responding to comments, and there may be some situations where the commenter is identified by name.
	<sup>51</sup> Officials from one program office within EPA responded that their office is not responsible for responding to public comments, but noted this responsibility is shared between program offices and a work group.

 $<sup>^{52}</sup>$ For participating agencies, a comment ID number is also known as a document ID. This is the unique identifier established for a document and includes the agency acronym, the year created, the docket, and the document number.
Officials from all program offices within CFPB and BLM responded to the survey that they never identified comments by commenter name, organization, or comment ID in their responses to public comments. In its response to comments in a 2014 final rule, for example, CFPB stated that "industry commenters also emphasized the need to coordinate with the States," without specifying the organization or specific comments. Similarly, in its response to comments document for a 2016 rule, for example, BLM responded directly to the themes and issues raised by comments while stating that the issue was raised by "one commenter" or "some commenters."

Selected Agencies' Practices Associated with Posting Identity Information Are Not Clearly Communicated to Public Users of Comment Websites

The 10 selected agencies have implemented varied ways of posting identity information during the comment intake process, particularly regarding posting duplicate comments, as allowed by the APA. Our analysis of Regulations.gov and agency-specific comment websites shows that these practices are not always documented or clearly communicated to public users of the websites. Public users are members of the public interested in participating in the rulemaking process via Regulations.gov or agency-specific websites. They may or may not submit a comment. In part to facilitate effective public participation in the rulemaking process, the E-Government Act requires that all public comments and other materials associated with the rulemaking docket should be made "publicly available online to the extent practicable." There may be situations where it is not practicable to post all submitted items, for example when resource constraints prevent the scanning and uploading of thousands of duplicate paper comments. Because the content of such comments is still reflected in the administrative record, such practices are not prohibited by the APA or the E-Government Act.

However, key practices for transparently reporting open government data state that federal government websites—like those used to facilitate the public comment process—should fully describe the data that are made available to the public, including by disclosing data sources and limitations.<sup>53</sup> This helps public users make informed decisions about how to use the data provided. In the case of identity information submitted with public comments, for example, public users may want to analyze identity information to better understand the geographic location from which comments are being submitted, and would need information about the

<sup>53</sup>GAO-19-72.

availability of address information to do so. The Administrative Conference of the United States has made several recommendations related to managing electronic rulemaking dockets.<sup>54</sup> These include recommendations that agencies disclose to the public their policies regarding the treatment of materials submitted to rulemaking dockets, such as those associated with protecting sensitive information submitted by the public.<sup>55</sup>

As described earlier in this report, the varied practices that selected agencies use with regard to identity information during the public comment process results in the inconsistent presentation of this information on the public websites, particularly when it is associated with duplicate comments. Although the APA and E-Government Act do not include any requirements associated with the collection or disclosure of identity information, we found that the selected agencies we reviewed do not effectively communicate the limitations and inconsistencies in how they post identity information associated with public comments. As a result, public users of the comment websites lack information related to data availability and limitations that could affect their ability to use the comment data and effectively participate in the rulemaking process themselves.

<sup>&</sup>lt;sup>54</sup>The Administrative Conference of the United States was established by statute in 1964 as an independent agency of the federal government. Its purpose is to promote improvements in the efficiency, adequacy, and fairness of the procedures by which federal agencies conduct regulatory programs, administer grants and benefits, and perform related governmental functions.

<sup>&</sup>lt;sup>55</sup>78 Fed. Reg. 41,352, 41,360 (July 10, 2013).

Selected Agencies' Practices Associated with Posting Identity Information on Regulations.gov Vary and Are Not Clearly Communicated to Public Users

Public users of Regulations.gov seeking to submit a comment are provided with a blanket disclosure statement related to how their identity information may be disclosed, and are generally directed to individual agency websites for additional detail about submitting comments. The Regulations.gov disclosure statements and additional agency-specific details are provided on the comment form, and a user seeking to review comments (rather than submit a comment) may not encounter them on Regulations.gov.<sup>56</sup> Regulations.gov provides the following disclosure statement at the bottom of each comment submission form:

Any information (e.g., personal or contact) you provide on this comment form or in an attachment may be publicly disclosed and searchable on the Internet and in a paper docket and will be provided to the Department or Agency issuing the notice. To view any additional information for submitting comments, such as anonymous or sensitive submissions, refer to the Privacy Notice and User Notice, the Federal Register notice on which you are commenting, and the Web site of the Department or Agency.

Similar information is provided to all public users in the Privacy Notice, User Notice, and Privacy Impact Assessment for Regulations.gov and the eRulemaking Program.<sup>57</sup> While all of these note that any information, personal or otherwise, submitted with comments may be publicly disclosed, public users are not provided any further detail on Regulations.gov regarding what information, including identity information, they should expect find in the comment data.

We found that when Regulations.gov provides public users with additional agency-specific information about the comment intake process, including accepting and posting comments, it is typically provided in the context of the comment form and does not provide public users enough detail to

<sup>57</sup>The Privacy Notice and User Notice disclose the ways in which Regulations.gov uses, discloses, and manages data associated with the website. The Privacy Impact Assessment is required by the E-Government Act of 2002, and provides public users with documented assurance that privacy issues associated with Regulations.gov have been identified and addressed.

<sup>&</sup>lt;sup>56</sup>Agency-specific details may also be provided in the text of the NPRM, but when public users of Regulations.gov extract or search comment data, the comment data are presented in a consistent format, so it is not intuitive for a user to expect that different posting processes would be followed. In recent years, there have been several efforts in academic research and the media to conduct large-scale analyses of comment data. For example, Steven J. Balla et al., *Where's the Spam? Mass Comment Campaigns in Agency Rulemaking*, working paper (The George Washington University Regulatory Studies Center: Apr. 2, 2018).

determine what comment data will be available for use when searching comments that are already submitted. Specifically, each comment form contains a pop-up box under the heading "Alternate Ways to Comment," which reflects the language associated with comment submission methods included in the NPRM on which individuals are seeking to comment. Additionally, three participating agencies in our review (EPA, FWS, and WHD) provide additional detail about posting practices on the comment form under the heading "Agency Posting Guidelines."<sup>58</sup> Both FWS and WHD indicate that the entire comment, including any identifying information, may be made available to the public. Although WHD follows DOL policy associated with posting duplicate comments, which allows some discretion in posting practices, according to a WHD official, without exception, all comments are posted to Regulations.gov. In our review of WHD comment data, we did not identify instances where this practice was not followed.

The "Agency Posting Guidelines" provided by EPA inform public users that all versions of duplicate or near-duplicate comments as part of mass mail campaigns may not be posted; rather a representative sample will be provided, with a tally of the total number of duplicate comments received. (See fig. 9.)

<sup>&</sup>lt;sup>58</sup>The other five participating agencies in the scope of this report do not include any information under the "Agency Posting Guidelines."





Source: GAO analysis of Regulations.gov information. | GAO-19-483

However, this information does not provide enough detail to help public users determine whether all of the individual comments and associated identity information are posted within this docket, because it indicates that samples are provided for duplicate comments, rather than all of the copies submitted. We found that one EPA docket received more than 350 separate sets of duplicate comments comprising a total of more than 4.3 million comments (as reported by Regulations.gov) but there is variation in how these comments were posted. Specifically, EPA inconsistently presented duplicate comments: 198 of the 350 duplicate comment sets in this docket were submitted via email. Of the duplicate comment sets

	submitted via email, 45 sets have all comments posted in Regulations.gov, while 153 sets have a sample of the comments posted. According to EPA officials, this inconsistency results from the format in which the comments were submitted. For example, when duplicate comments are compiled into a single document and submitted to EPA through one email, all of the comments will be posted, whereas duplicate comments that are emailed separately will be accounted for in the tally accompanying a sample comment.
	While the APA and the E-Government Act do not require comments to be posted in any particular way, EPA has established detailed internal guidance for the comment intake process for its Docket Center staff. This document is in draft form, but clearly lays out the processes EPA staff are expected to follow when duplicate comments are submitted in different ways, and what naming conventions will be used in different instances. However, EPA does not provide similar information to public users about the process it uses to determine whether all duplicate comments will be posted, making it challenging for public users to determine whether all comments are available on Regulations.gov. <sup>59</sup>
Participating Agency Websites	The eRulemaking PMO provides participating agencies with flexibility in how they choose to use FDMS and Regulations.gov, with each department or agency responsible for managing its own data within the website. As a result, Regulations.gov directs public users to participating agencies' websites for additional information about agency-specific review and posting policies. We found that all of the selected participating agencies provide additional information of some kind about the public comment process on their own websites. However, the provided information usually directs users back to Regulations.gov or to the <i>Federal Register</i> . Further, even when selected participating agencies include details on their website about the agency's posting practices or treatment of identity information associated with public comments, it does not fully describe data limitations that public users need to make informed decisions about how to use the data provided.

<sup>&</sup>lt;sup>59</sup>This challenge is not limited to the general public. To compile, review, and prepare a summary report related to the mass mail comments submitted to the docket discussed in this section, EPA officials reported that it took them approximately 55 hours. The compiled information included whether the comments were included in FDMS or not, the location of comments stored outside of FDMS, and the format of the comment submissions for each mass mail campaign.

Specifically, seven of the eight participating agencies (BLM, CMS, CFPB, EPA, FWS, FDA, and WHD) direct public users back to Regulations.gov and the *Federal Register*, either on webpages that are about the public comment process in general, or on pages containing information about specific NPRMs. As discussed previously, however, the disclosure statement on Regulations.gov directs public users to the agency website for additional information. Although three of these participating agencies (EPA, FWS, and FDA) do provide public users with information beyond directing them back to Regulations.gov or the *Federal Register*, only FDA provides users with details about posting practices that are not also made available on Regulations.gov.

- EPA: The additional information provided on EPA's website largely replicates the "Agency Posting Guidelines" provided on the Regulations.gov comment form, as shown in figure 9. As discussed previously, however, the way in which EPA posts duplicate comments varies, and the provided information does not include details about the process the agency uses to determine whether all duplicate comments will be posted.
- FWS: One NPRM-specific web page that we identified communicated to public users that all comments will be posted on Regulations.gov, including any personal information provided through the process. This largely replicates the "Agency Posting Guidelines" provided on the Regulations.gov comment form, as well as language included in the NPRM itself. However, according to an FWS official, when the agency receives hard-copy duplicate comments through the mail, only one sample of the duplicate is posted publicly on Regulations.gov. FWS does not have any policies related to this practice and the information FWS provides to public users does not include details about how the agency determines which comment to post as the sample.
- **FDA:** On its general website, FDA includes a webpage titled, "Posting of Comments." On this page, FDA provides users with a detailed explanation about a policy change the agency made in 2015 related to the posting of public comments submitted to rulemaking proceedings. Specifically, prior to October 2015, FDA did not publicly post comments submitted by individuals in their individual capacity. See figure 10.



#### Figure 10: Example Food and Drug Administration Explanation for a Comment Withheld from Public Posting

Source: GAO analysis of Regulations.gov information. | GAO-19-483

Note: Metadata summarize and describe the data contained in a dataset.

After October 15, 2015, FDA's policy is to publicly post all comments to Regulations.gov, to include any identifying information submitted with the comment.<sup>60</sup> In our review of FDA comments submitted to dockets opened since October 15, 2015, we did not identify instances where this policy was not followed.

The one participating agency in our scope (EBSA) that does not direct public users back to Regulations.gov instead recreates the entire rulemaking docket on its own website. On the main EBSA webpage related to regulations, public users can find links to various websites related to rulemaking, including a "Public Comments" page, but not Regulations.gov.<sup>61</sup> From the "Public Comments" page, public users can access pages that are specific to NPRMs and other activities for which

<sup>61</sup>Public users are directed to other webpages within the EBSA site, the *Federal Register*, the Electronic Code of Federal Regulations, and Reginfo.gov, which displays regulatory actions and information collections under review at the Office of Information and Regulatory Affairs within the Office of Management and Budget.

<sup>&</sup>lt;sup>60</sup>FDA provides an exception for commenters wishing to submit a comment containing confidential information. To do so, a commenter must submit a comment marked "confidential" by mail or hard-copy delivery, and also provide a redacted version suitable for public posting.

EBSA is requesting public comments.<sup>62</sup> On the NPRM-specific webpages, the rulemaking docket that can be found on Regulations.gov is duplicated, including individual links to each submitted comment.63 Certain document links, such as those for the proposed rule or final rule, direct a public user to the Federal Register document, but the comment links do not direct users to Regulations.gov.<sup>64</sup> While EBSA follows DOL guidance associated with posting duplicate comments, which allows some discretion in posting practices, EBSA does not have a policy for how comments are posted to Regulations.gov or its own website, and in the examples we reviewed the content of the docket pages does not alwavs match. According to EBSA officials, the agency began this practice prior to the development of Regulations.gov, and has continued it because internal staff and other stakeholders find the webpages useful. However, we have previously reported that reducing or eliminating duplicative government activities can help agencies provide more efficient and effective services.65

Further, on EBSA's "Public Comments" webpage, public users are informed that comments with inappropriate content will be removed, but no other information associated with EBSA's posting practices is provided on this general page. In one instance on an NPRM-specific webpage, public users are informed that identity information has been removed from certain comments due to the inclusion of personal health information, but most of the NPRM-specific webpages we reviewed did not include this disclosure. Additionally, duplicate comments are posted on the NPRMspecific webpages under the heading "Petitions," and are posted with a number following the title of the comment. While public users are informed that the number represents the total number of comments submitted, not all links include a copy of each individual comment. This practice aligns with DOL guidance, but as a result, the way in which

<sup>64</sup>The *Federal Register* does include links to Regulations.gov.

<sup>65</sup>Duplication occurs when two or more agencies or programs are engaged in the same activities or provide the same services to the same beneficiaries. See GAO's Duplication and Cost Savings web page: http://www.gao.gov/duplication/overview.

<sup>&</sup>lt;sup>62</sup>These include requests for information, which are not subject to APA Notice and Comment requirements.

<sup>&</sup>lt;sup>63</sup>Certain elements available on Regulations.gov are not readily available on the EBSA NPRM-specific page without clicking on additional documents. These include items such as relevant dates, Regulatory Information Number, and Code of Federal Regulations citations.

EBSA posts duplicate comments varies even within dockets, and the provided information does not include details about the process the agency uses to determine whether all duplicate comments will be posted. Additionally, because EBSA recreates rulemaking dockets on its own website without referencing Regulations.gov or explaining the process, public users lack assurance about how EBSA's data sources relate to one another.

Because participating agencies are not required to adhere to standardized posting practices, Regulations.gov directs public users to participating agency websites for additional information about posting practices and potential data limitations. However, the additional information provided on the selected agencies' websites is rarely different from what is provided on Regulations.gov. Further, it does not describe the limitations associated with the identity information contained in publicly posted comments, and in many cases simply directs users back to Regulations.gov. As allowed for under the APA, all of the participating agencies in our review vary in the way in which they post identity information associated with comments—particularly duplicate comments. However, the lack of accompanying disclosures may potentially lead users to assume, for example, that only one entity has weighed in on an issue when, actually, that comment represents 500 comments.

The APA, E-Government Act and relevant Executive Orders establish the importance of public participation in the rulemaking process, to include access to electronic rulemaking dockets in formats that can be easily searched and downloaded. Further, key practices for transparently reporting open government data state that federal government websites—like those used to facilitate the public comment process—should fully describe the data that are made available to the public, including by disclosing data sources and limitations.<sup>66</sup> Without better information about the posting process, the inconsistency in the way in which duplicate comments are presented to public users of Regulations.gov limits public users' ability to explore and use the data and could lead users to draw inaccurate conclusions about the public comments that were submitted and how agencies considered them during the rulemaking process.

<sup>66</sup>GAO-19-72.

#### Agency-Specific Comment Websites Do Not Clearly Communicate Posting Policies to Public Users

Both SEC and FCC use comment systems other than Regulations.gov and follow standardized posting processes associated with public comments submitted to their respective comment systems, but SEC has not clearly communicated these practices to the public. Although it appears to users of the SEC website that the agency follows a consistent process for posting duplicate comments, this practice has not been documented or communicated to public users of its website. As discussed earlier, SEC posts a single example for each set of duplicate comments and indicates the total number of comments received. As a result, the identity information and any unique comment content beyond the first example are not accessible to the public online. According to SEC officials, this practice is not documented in formal policy, and is not explicitly communicated to public users of the SEC's comment website. Although SEC does provide public users with some information on its "How to Submit Comments" page, this information is limited to informing public users that all comments will be posted publicly, without any edits to personal identifying information, and no other information related to SEC's posting process is provided. Without clearly communicated policies for posting comments, public users of SEC.gov do not have information related to data sources and limitations needed to determine whether and how they can use the data associated with public comments.

In contrast, FCC identifies its policies for posting comments and their associated identity information in a number of places on the FCC.gov website, and on the ECFS web page within the general website. Regarding comments submitted to rulemaking proceedings through ECFS, public users are informed that all information submitted with comments, including identity information, will be made public.<sup>67</sup> According to FCC officials, all comments are posted directly to ECFS as they are submitted, without intervention by FCC staff. Further, according to officials, all duplicate comments remain in ECFS as individual comments, unless an organization submits a Standard filing with an attached file containing multiple comments. Our review of ECFS comment data did not identify discrepancies with this practice.

<sup>&</sup>lt;sup>67</sup>In addition to posting policies associated with rulemaking proceedings, FCC also provides public users with information associated with other types of comments, such as submissions to FCC's consumer complaint database or comments made on FCC's various social media. For example, guidance provided on FCC's "Comment Policy" web page is specific to informal comments submitted to FCC online, and although it indicates that certain types of comments will be removed from posting, public users are also clearly informed these types of comments are not a substitute for formal submissions to ECFS.

#### Conclusions

While the public comment process allows interested parties to state their views about prospective rules, the lack of communication with the public about the way in which agencies treat identity information during the posting process, particularly for duplicate comments, may inhibit users' meaningful participation in the rulemaking process. While the APA does not include requirements for commenters to provide identity information, or for agency officials to include commenter identity as part of their consideration of comments, key practices for transparently reporting open government data state that federal government websites—like those used to facilitate the public comment process—should fully describe the publicly available data, to include disclosing data sources and limitations. Without clearly communicating how comments and their associated identity information are presented in the data, public users could draw inaccurate conclusions about public comments during the rulemaking process, limiting their ability to participate in the rulemaking process.

Five selected agencies do not have a policy for posting comments, and the selected agencies generally do not clearly communicate to public users about the way in which they publicly post comments and their associated identity information. In addition, one agency fully duplicates rulemaking dockets on its own website, without informing users that the information may be found in a searchable database on Regulations.gov. Regulations gov does not provide detailed information about posting policies, and seven of the eight participating agencies in the scope of our review direct public users back to Regulations.gov or the Federal Register on their own websites. Further, the available information is provided on the comment form, so public users seeking to review comment data that had been previously submitted may not encounter it. Because all of the participating agencies in our review vary in the way in which they post identity information associated with comments-particularly duplicate comments-the lack of accompanying disclosures may potentially lead users to reach inaccurate conclusions about who submitted a particular comment, or how many individuals weighed in on an issue. As a result, public users of Regulations.gov do not have information related to data sources and limitations that could affect their ability to effectively use the comment data and, consequently, participate in the rulemaking process. Similarly, users of SEC.gov do not have information related to data sources and limitations needed to determine whether and how they can use the data associated with public comments, because the agency lacks a policy for posting duplicate comments and associated identity information to the public. In short, more clearly communicated information about posting policies, particularly with regard to identity information and duplicate comments, could help public users make informed decisions

	about how to use the comment data these agencies provide, and how comments may have informed the rulemaking process.	
Recommendations for Executive Action	We are making the following eight recommendations to the Directors of BLM, CFPB, and FWS; the Administrators of CMS, EPA, and WHD; the Assistant Secretary of Labor for EBSA; and the Chairman of the SEC, respectively:	
	The Director of BLM should create and implement a policy for standard posting requirements regarding comments and their identity information, particularly for duplicate comments, and should clearly communicate this policy to the public on the BLM website. (Recommendation 1)	
	The Administrator of CMS should create and implement a policy for standard posting requirements regarding comments and their identity information, particularly for duplicate comments, and should clearly communicate this policy to the public on the CMS website. (Recommendation 2)	
	The Director of CFPB should finalize its draft policy for posting comments and their identity information, particularly for duplicate comments, and clearly communicate it to the public on the CFPB website. (Recommendation 3)	
	The Assistant Secretary of Labor for EBSA should	
	<ol> <li>create and implement a policy for standard posting requirements regarding comments and their identity information, particularly for duplicate comments;</li> </ol>	
	<ol><li>clearly communicate this policy to the public on the EBSA website; and</li></ol>	
	3. evaluate the duplicative practice of replicating rulemaking dockets on the EBSA website, to either discontinue the practice or include a reference to Regulations.gov and explanation of how the pages relate to one another. (Recommendation 4)	
	The Administrator of EPA should finalize its draft policy for posting comments and their identity information, particularly for duplicate comments, and clearly communicate it to the public on the EPA website. (Recommendation 5)	

	The Director of FWS should create and implement a policy for standard posting requirements regarding comments and their identity information, particularly for duplicate comments, and should clearly communicate this policy to the public on the FWS website. (Recommendation 6)
	The Chairman of the SEC should develop a policy for posting duplicate comments and associated identity information and clearly communicate it to the public on the SEC website. (Recommendation 7)
	The Administrator of WHD should clearly communicate its policy for posting comments and their identity information, particularly for duplicate comments, to the public on the WHD website. (Recommendation 8)
Agency Comments and Our Evaluation	We provided drafts of this product for comment to CFPB, EPA, FCC, SEC, the Department of Health and Human Services, the Department of the Interior, and DOL. We received written comments from three of the selected agencies and the three Departments which are reproduced in appendixes V through X. All of the selected agencies generally agreed with the recommendations directed to them and indicated that they intended to take action to more clearly communicate their posting policies to the public. BLM, EBSA, FWS, and SEC also stated that they intend to develop written policies associated with posting comments.
	In its written comments, the Department of Health and Human Services stated that CMS already has policies for standard posting requirements. However, CMS could not provide us with this policy during the course of our review, and in the accompanying technical comments, officials stated that guidance associated with posting comments has not been formalized in a written document. Given that we found significant variation in the way that CMS posts comments, even within a single docket, we continue to believe that it is important for CMS to develop and implement a standard policy for posting comments and their identity information, in addition to communicating this policy to the public on the CMS website.
	CFPB and EPA also stated that they intend to finalize their draft policies for posting comments and their associated identity information. In addition, EPA included technical comments in its letter, which we considered and incorporated in this report as appropriate. FCC had no comments on the draft report, but provided technical comments, which we incorporated as appropriate. The remaining selected agencies and departments also provided technical comments, which we considered and incorporated in this report as appropriate.

As arranged with your offices, unless you publicly announce the contents of this report earlier, we plan no further distribution until 30 days from the report date. At that time, we will send copies of this report to the appropriate congressional committees; the Director of CFPB; the Administrator of EPA; the Chairmen of FCC and SEC; and the Secretaries of Health and Human Services, the Interior, and Labor. In addition, the report will be available at no charge on GAO's website at http://www.gao.gov.

If you or your staff have any questions about this report, please contact me at (202) 512-6722 or bagdoyans@gao.gov. Contact points for our Offices of Congressional Relations and Public Affairs may be found on the last page of this report. GAO staff who made key contributions to this report are listed in appendix XI.

Seto J. B

Seto J. Bagdoyan, Director of Audits Forensic Audits and Investigative Service

#### List of Requesters

The Honorable Rob Portman Chairman The Honorable Thomas R. Carper Ranking Member Permanent Subcommittee on Investigations Committee on Homeland Security and Governmental Affairs United States Senate

The Honorable James Lankford Chairman Subcommittee on Regulatory Affairs and Federal Management Committee on Homeland Security and Governmental Affairs United States Senate

The Honorable Robert C. "Bobby" Scott Chairman Committee on Education and Labor House of Representatives

The Honorable Frank Pallone, Jr. Chairman Committee on Energy and Commerce House of Representatives

The Honorable Elijah E. Cummings Chairman The Honorable Jim Jordan Ranking Member Committee on Oversight and Reform House of Representatives

The Honorable Gerald E. Connolly Chairman Subcommittee on Government Operations Committee on Oversight and Reform House of Representatives The Honorable Yvette D. Clarke House of Representatives

The Honorable Eliot L. Engel House of Representatives

The Honorable Hakeem S. Jeffries House of Representatives

The Honorable Carolyn B. Maloney House of Representatives

The Honorable Gregory W. Meeks House of Representatives

The Honorable Paul D. Tonko House of Representatives

The Honorable Nydia M. Velázquez House of Representatives

### Appendix I: Survey of Program Offices with Regulatory Responsibilities within Selected Agencies

To determine how selected agencies treat identity information associated with public comments, in October 2018 we surveyed and received responses from 52 program offices within the selected agencies about their practices associated with comment intake (including identifying duplicate comments and posting comments to the public website), comment analysis (including reviewing comments and considering their content), and response to comments. To select the program offices to receive survey questionnaires about the public comment process, we first reviewed agency websites to identify all of the program offices in each of the selected agencies. We then identified program offices with regulatory responsibilities described by the websites and that had issued at least one Notice of Proposed Rulemaking (NPRM) from 2013 through 2017, and provided these lists to the selected agencies for confirmation. Table 7 lists the program offices we surveyed.

Table 7: Program Offices within Selected Agencies with Regulatory Responsibilities

Bureau of Land Management (BLM), Department of the Interior

- 1. Assistant Director for Resources and Planning
- 2. Assistant Director for Energy, Minerals, and Realty Management

Centers for Medicare & Medicaid Services (CMS), Department of Health and Human Services

- 3. Center for Medicaid and Children's Health Insurance Program Services
- 4. Center for Consumer Information and Insurance Oversight
- 5. Center for Medicare—Parts C and D
- 6. Office of Financial Management
- 7. Center for Clinical Standards and Quality
- 8. Center for Medicare and Medicaid Innovation
- 9. Center for Program Integrity
- 10. Office of Information Technology
- 11. Center for Medicare—Fee for Service

Consumer Financial Protection Bureau (CFPB)

12. Office of Regulations

Employee Benefits Security Administration (EBSA), Department of Labor

- 13. Office of Exemptions Determinations
- 14. Office of Health Plan Standards and Compliance Assistance
- 15. Office of Regulations and Interpretations
- 16. Office of Policy and Research

Environmental Protection Agency (EPA)
17. Office of Air and Radiation
18. Office of Chemical Safety and Pollution Prevention
19. Office of Land and Emergency Management
20. Office of Water
Federal Communications Commission (FCC)
21. Consumer and Governmental Affairs Bureau
22. Office of Engineering and Technology
23. International Bureau
24. Media Bureau
25. Public Safety and Homeland Security Bureau
26. Wireless Telecommunications Bureau
27. Wireline Competition Bureau
Fish and Wildlife Service (FWS), Department of the Interior
28. National Wildlife Refuge System
29. Fish and Aquatic Conservation
30. Ecological Services
31. Migratory Birds
32. International Affairs
33. Wildlife and Sport Fish Restoration Programs
34. Chief Law Enforcement
35. Center for Biologics Evaluation and Research
36. Center for Drug Evaluation and Research
37. Center for Devices and Radiological Health
38. Center for Food Safety and Applied Nutrition
39. Center for Tobacco Products
40. Center for Veterinary Medicine
41. Office of Chief Counsel
42. Office of Combination Products
43. Office of Policy
44. Office of Regulatory Affairs
Securities and Exchange Commission (SEC)
45. Division of Corporation Finance
46. Division of Economic and Risk Analysis
47. Division of Investment Management
48. Division of Trading and Markets

50. Office of the Chief Acc	
JU. UNICE UI UIE UNIEI ACC	ountant
51. Office of Credit Rating	3
Wage and Hour Division	(WHD), Department of Labor
52. Assistant Administrato	r for the Office of Policy

Survey Development	We developed a draft survey questionnaire in conjunction with another GAO engagement team conducting work on the public comment process, and pretested it with program office officials from four of the selected agencies in August and September 2018. We interviewed these officials to improve the questionnaire and ensure that (1) the questions were clear and unbiased, (2) the information could be feasibly obtained by program office officials, (3) the response options were appropriate and reasonable, and (4) the survey did not create an undue burden on program office officials. The process of developing the survey was iterative, where we used the results of one pretest to modify the questionnaire for the next pretest.
Survey Administration and Review	We distributed the questionnaires to the program offices as fillable Portable Document Format (PDF) forms, in October 2018 requesting that officials collaborate with others in their office to ensure the responses were reflective of the program office as a whole, rather than one individual's experience. Two agencies, CMS and SEC, have agency-level administrative offices with centralized responsibilities for certain aspects of the public comment process. For these agencies, the selected program offices were instructed to leave certain questions blank, and we provided separate questionnaires for the administrative offices. All 52 program offices completed the survey, but the results cannot be generalized to program offices outside of the selected agencies.
	In developing, administering, and analyzing this survey, we took steps to minimize the potential errors that may result from the practical difficulties of conducting any survey. Because we surveyed and received responses from all program offices with regulatory responsibilities in the selected agencies, our results are not subject to sampling or nonresponse error. We pretested and reviewed our questionnaire to minimize measurement error that can arise from differences in how questions are interpreted and the sources of information available to respondents. We also answered

	questions from program offices during the survey, reviewed completed questionnaires, and conducted follow-up as necessary. On the basis of this follow-up and with agreement from the responding officials, we edited responses as needed. For CMS and SEC, we edited the blank questions in the program office questionnaires with responses from their administrative offices.
Relevant Survey Questions	<ul> <li>Information collected from the survey we conducted will also be used in other forthcoming GAO reports that are focused on the public comment process. The specific questions and response options from the survey that were analyzed in this report are reproduced below. See the body of the report for the results. Other questions included in the survey will be reproduced in the reports that include their results.</li> <li>3. Approximately how many rulemaking proceedings does your office solicit comments on in a typical year? Please only include rulemaking proceedings subject to notice-and-comment under the APA. (Click one button)</li> </ul>
	<ul> <li>0</li> <li>1-5</li> <li>6-10</li> <li>11-15</li> <li>16-20</li> <li>More than 20</li> <li>4. Considering the rulemaking proceedings since 2013 your office has been responsible for that had a public comment period, approximately how many comments were submitted to the one rulemaking that received the most total comments during that time period?</li> <li>1-500</li> <li>501-1500</li> </ul>
	<ul> <li>1501-5,000</li> <li>5,001-25,000</li> <li>25,001-100,000</li> <li>100,001-500,000</li> </ul>

- 500,001-1,000,000
- More than 1,000,000

5. f. Considering the rulemaking proceedings since 2013 your office has been responsible for that had a public comment period, in approximately how many rulemakings have comments been submitted anonymously?

- All or almost all rulemakings
- Most rulemakings
- About half of rulemakings
- Some rulemakings
- No rulemakings
- Don't know

7. Is your program office responsible, in whole or in part, for managing the intake of public comments submitted to federal rulemaking proceedings subject to notice-and-comment under the APA?

- Yes
- No please identify the responsible office(s) and contact them as necessary to complete the following questions:

10. Is your program office responsible, in whole or in part, for identifying duplicative comments (carrying out a de-duplication process)?

- Yes
- No please identify the responsible office(s) and contact them as necessary to complete the following questions:

13. Is your program office responsible, in whole or in part, for analyzing public comments submitted to federal rulemaking proceedings subject to notice-and-comment under the APA?

- Yes
- No please identify the responsible office(s) and contact them as necessary to complete the following questions:

13.b(c). Considering the rulemaking proceedings since 2013, how important to your analysis, if at all, is the identity of the individual commenter?

- Extremely important
- Very important
- Moderately important
- Slightly important
- Not at all important
- Don't know

13.b(d). Considering the rulemaking proceedings since 2013, how important to your analysis, if at all, is the organizational affiliation, if any, of the commenter?

- Extremely important
- Very important
- Moderately important
- Slightly important
- Not at all important
- Don't know

16. Is your program office responsible, in whole or in part, for responding to public comments submitted to federal rulemaking proceedings subject to notice-and-comment under the APA?

- Yes
- No please identify the responsible office(s) and contact them as necessary to complete the following questions:

16.b.(a). Considering the public responses to comments that your office has drafted for rulemaking proceedings since 2013, how frequently did the responses identify any specific comments by commenter name, organization, or comment ID?

- All or almost all responses
- Most responses
- About half of responses
- Some responses
- No responses
- Don't know

### Appendix II: Regulations.gov Comment Form Example

Comments are submitted to Regulations.gov via an electronic comment form. See figure 11 for an example of a comment form from Regulations.gov.

ocket; R or related	and Drug Administration (FDA) Notice: Post-Marketin Request for Comments d information, Open Docket Folder 🔁	g Pediatric-Focused Product Safety Reviews; Establishment of a Publi
or related	d information, Open Docket Folder 🔁	
G	Your Information	
G	Your Information	
		🚯 Your Information 🍃 🛛 Your Preview 🍃 🤂 Your Receip
	Information entered will be viewable on Regulations.go	v View Commenter's Checklist (PDF)   Alternate Ways to Comm
C	Comment (Required) 🔇	
ſ		
		5000 characters remaining
U	Jpload file(s) (Optional) 🔇	- Flag have
1	Choose files	op files here
	610030 1103	
-	irst Name Last Name	
E E	irst Name Last Name	
Г	I want to provide my contact information	
Ē	I am submitting on behalf of a third party	
C	Category (Required) 🔇	
9	Select a Category +	
		Continue
ease see	e the Privacy Notice and User Notice regarding commer	nt submission.
		ment form or in an attachment may be publicly disclosed and searchable o

Source: Regulations.gov. | GAO-19-483

### Appendix III: Electronic Comment Filing System Comment Forms

The Federal Communications Commission's (FCC) Electronic Comment Filing System (ECFS) allows commenters to submit comments to rulemaking proceedings via a Standard filing and Express filing.<sup>1</sup> A Standard filing allows commenters to attach a file to their comment. See figure 12 for an example of a Standard filing.

### Figure 12: Federal Communications Commission Electronic Comment Filing System's Standard Filing Comment Form

Submit a Filir		_
Proceeding(s)	* Required - Type to snarch proceedings. Press SNTER key after each selection	
Name(s) of Filer(s)	*Required - Add File(s), Press INTER law other each entry	
Law Firm(s)	Add Lawfernish Press INTER key after each entry	
Attorney/Author Name(s)	Add Author(s). Press ENTER key after each entry	
Primary Contact Email	example@email.com	
Type of Filing		
File Number		
Report Number		
Bureau ID Number		
Address of	* Required	
Address	* Required	
Address 2		
City	* Required	
State	* Required	
ZIP	* Required	
Upload Documents	Click or drop files here: PDF, toot, ppt, pdt, door, sha, doo, sha, fde, yd, dog (oerend files) may see: 25 MB per submixing (may number of files 5)	

Source: FCC.gov. | GAO-19-483

<sup>&</sup>lt;sup>1</sup>FCC's ECFS also accepts comments in response to specific types of FCC actions via Non-Docketed filings. Users are instructed not to use Non-Docketed filings to submit comments in a proceeding for which a docket number or rulemaking number has been assigned.

An Express filing does not allow for files to be attached. See figure 13 for an example of an Express filing.

Figure 13: Federal Communications Commission Elec	ctronic Comment Filing
System's Express Filing Web Form	-

Federal Communications Commission	Fills	ngs & Proceedings SEARCH F	Submit a Check filing ILING STATUS	User HELP
	Standard Filing	Express Comment	Non-Docketed Filing	
ECFS Express		COMMENT REVIEW	) (3) W CONFIRMATION	
Proceeding(s)	* Required - Type t	to search proceedings. Pre	ss ENTER key after each selection	
Name(s) of Filer(s)	*Required - Add Fi	ler(s). Press ENTER key afte	er each entry	
Primary Contact Email	example@email.cor	example@ernail.com		
Address	* Required			
Address 2				
City	* Required			
State	* Required			
ZIP	* Required			
Brief Comments				
	Note: You are filing a will be publicly available	a document into an official able via the web.	FCC proceeding. All information sub	omitted, including names and addresses,
	Email Confirmation			

Source: FCC.gov. | GAO-19-483

### Appendix IV: Securities and Exchange Commission Comment Form Example

One way in which comments are submitted to the Securities and Exchange Commission (SEC) is through an electronic comment form. See figure 14 for an example of a comment form from SEC.gov.

Proposed Revision	Submit Comments on 57-30-18 s to Prohibitions and Restrictions on Proprietary Trading and Certain Interests in, and Relationships Wi Hedge Funds and Private Equity Funds
	Complete this form to submit your comments. Fields marked with an asterisk (*) are required.
	Important: All comments will be made available to the public. Submit only information that you wish to make available publicly. This form should not be used for submitting a tip or complaint (including a whistleblower complaint). If you wish to submit a tip or complaint regarding a potential violation of the federal securities laws, please use our <u>onlina form</u> , or mail your information to the SEC complaint Center at the following address: SEC Complaint Center, 100 F Street NE, Washington, D.C. 20549-0213. You can also send a fax to 703-613-6965.
	If you would like to submit Microsoft Word, PDF, or WordPerfect files as your comments or as supplementary materials to your comments, you will be able to do so on the next page.
* First Name	
Middle Initial	
* Last Name	
Suffix (if any)	
	examples:
	PhD. CPA Exure
Professional Affiliation	
(if any)	examples: Professor of Economics, University of Providence (or) Markets Section Char, Association of Professional Accountants (or) CED, Anner Welger Company
City	
State / Province	×
Country	United States V
* Email address	
Your email address will not be published on the web site	example. nameĝdomalmame domaln
* Comments:	
If you are submitting your comment in an attached file, put the words "Comments attached" in the text box. You will be able to attach up to 3 documents after you click "continue."	
	~
	Continue
ttps://www.sec.gov/cgi-	bin/ruling-comments

#### Figure 14: Securities and Exchange Commission Comment Form Example

Source: SEC.gov. | GAO-19-483

## Appendix V: Agency Comments from the Bureau of Consumer Financial Protection



The Bureau looks forward to working with GAO as it monitors the Bureau's progress in implementing this recommendation. Sincerely, Kathleen Kraninger Kathleen L. Kraninger Director consumerfinance.gov 2

## Appendix VI: Agency Comments from the Environmental Protection Agency





associated with each mass mail campaign. that determining whether the comments were stored in FDMS or not took them approximately 55 hours to complete." Finally, it is important to note that the docket referenced in the report is the largest EPA docket ever in terms of number of comments received. In 2014, this docket received 4,383,712 public comments, out of the 6,051,473 total comments EPA received that year. To put that into perspective, in 2013, EPA received just 2,196,931 comments in total. To provide this context, EPA recommends the following change (in red italics) on page 37: "We found that one EPA docket, the Agency's largest docket in terms of comment count to date, received more than 350 separate sets of duplicate comments comprising a total of more than 4.3 million comments (as reported by Regulations.gov) .... " We appreciate the opportunity to review and respond to the draft report. If you require additional information or would like to discuss further, please contact Patricia Williams (202) 564-0204. Sincerely, Eina Vaughn Noga For Chief Information Officer and Deputy Assistant Administrator for Environmental Information Erin Collard, OMS cc: Patricia Williams, OMS Janice Jablonski, OMS Juanita Standifer, OMS Rebecca Moser, OMS Courtney Kerwin, OMS Pam Shenefiel, OMS Annette Morant, OCFO Dave Bruno, GAO Elizabeth Kowalewski, GAO

## Appendix VII: Agency Comments from the Department of Health and Human Services

o COMPANSIANCES (2)			
	DEPARTMENT OF HEALTH & HUMAN SERVICES	OFFICE OF THE SECRETARY	
to in manage		Assistant Secretary for Legislation Washington, DC 20201	
JUN	0 7 2019		
Seto Bagdoya	an		
Director, For	ensic Audits and Investigative Service nent Accountability Office		
441 G Street Washington,	NW		
Dear Mr. Bag			
"Federal Rule	comments on the U.S. Government Accountable emaking: Selected Agencies Should Clearly Co Information in the Public Comment Process" (	mmunicate Practices Associated	
The Departme	ent appreciates the opportunity to review this re	eport prior to publication.	
	Sincerely,		
	8 C.	an for	
	Matthew D. Ba Assistant Secre	ssett tary for Legislation	
Attachment			

9 9	
CMS already has policies for standard posting requirements regarding comments and their identity information, including for duplicate comments. CMS will communicate these policies to the public on the CMS website.	
HHS Response HHS concurs with this recommendation.	
<b><u>Recommendation 1</u></b> The Administrator of CMS should create and implement a policy for standard posting requirements regarding comments and their identity information, particularly for duplicate comments, and should clearly communicate this policy to the public on the CMS website.	
HHS appreciates GAO's review, and our response to the recommendation is as follows.	
The Administrative Procedure Act does not require commenters to provide identity information when submitting public comments. However, if identity information is provided, CMS policy is to post it with the comment. In addition, if multiple identical comments are received (such as form letters), agencies have discretion under the Administrative Procedure Act regarding how to post and identify these comments. CMS often receives duplicate comments or form letters, and our current policy is to post each comment from a unique commenter individually.	
HHS is committed to maintaining public trust in the regulatory process through transparency and consistency in our notice and comment procedures. The rulemaking process is governed by the Administrative Procedure Act and managed government-wide by the Office of Management and Budget. The Centers for Medicare & Medicaid Services (CMS) has internal policies regarding public posting requirements, including regarding identity information and duplicate comments, in accordance with the Administrative Procedure Act.	
The U.S. Department of Health & Human Services (HHS) appreciates the opportunity from the Government Accountability Office (GAO) to review and comment on this draft report.	
SERVICES ON THE GOVERNMENT ACCOUNTABILITY OFFICE'S DRAFT REPORT ENTITLED - FEDERAL RULEMAKING: SELECTED AGENCIES SHOULD CLEARLY COMMUNICATE PRACTICES ASSOCIATED WITH IDENTITY INFORMATION IN THE PUBLIC COMMENT PROCESS (GAO-19-483)	

## Appendix VIII: Agency Comments from the Department of the Interior

Conted S	tates Department of the Interior OFFICE OF THE SECRETARY Washington, DC 20240
	JUN 1 4 2019
Seto J. Bagdoyan Director, Audits	
Forensic Audits and Investigativ U.S. Government Accountabilit	
441 G Street NW Washington, DC 20548	y once
Dear Mr. Bagdoyan:	
summary of actions the Bureau Service (FWS) have planned to Recommendation 1: The Dira standard posting requirement	at two recommendations to address its findings. Below is a of Land Management (BLM) and the U.S. Fish and Wildlife implement the recommendations. ector of BLM should create and implement a policy for as regarding comments and their identity information, ments, and should clearly communicate this policy to the
regarding public comments and	will develop and issue policy for standard posting requirements their identity information. The policy will also include how to d the requirement for posting the policy on BLM's website.
standard posting requirement	ector of FWS should create and implement a policy for as regarding comments and their identity information, aments, and should clearly communicate this policy to the
regarding public comments and	will develop and issue policy for standard posting requirements their identity information. The policy will also include how to d the requirement for posting the policy on FWS' website.

10.12 If you have any questions about this response, or need additional information, please contact Nancy Thomas at (202 208-7954. Sincerely, Covern Au Scott J. Cameron Principal Deputy Assistant Secretary for Policy, Management and Budget

## Appendix IX: Agency Comments from the Department of Labor

U.S. Department of Labor	Office of the Assistant Secretary for Policy Washington, D.C. 20210	
Seto J. Bagdoyan Director Forensic Audits and Investigati U.S. Government Accountabili 441 G. Street, N.W. Washington, D.C. 20548		
Dear Mr. Bagdoyan:		
Office (GAO) report entitled, "	(DOL) is in receipt of the draft Gov Selected Agencies Should Clearly C nation in the Public Comment Proce	Communicate Practices
	nt recommendations. Two of those nort both recommendations. Responsent.	
If you would like additional inf 5080.	ormation, or have any questions ple	ase contact me at 202-693-
Sincerely,		
Stephanie Swirsky		
Deputy Assistant Secretary for	Policy	
Enclosures		

GAO draft report entitled, "Selected Agencies Should Clearly Communicate Practices Associated with Identity Information in the Public Comment Process" (GAO-19-483, Job Code 102648)
DOL's Response to GAO's Recommendations for Executive Action
Recommendation 4
<ul> <li>The Assistant Secretary of Labor for EBSA should: <ul> <li>(1) Create and implement a policy for standard posting requirements regarding comments and their identity information, particularly for duplicate comments;</li> <li>(2) Clearly communicate this policy to the public on the EBSA website; and</li> <li>(3) Evaluate the duplicate practice of replicating rulemaking dockets on the EBSA website, to either discontinue the practice or include a reference to Regulations.gov and explanation of how the pages relate to one another.</li> </ul> </li> </ul>
<ul> <li>EBSA response:         <ul> <li>(1) Agreed. EBSA will examine any such written policies of other DOL agencies, including DOL's Wage and Hour Division, and develop a written policy or policies regarding the posting of comments, including those with identical or near-identical comment text but with unique identity information (i.e., duplicate comments).</li> </ul> </li> </ul>
(2) Agreed. EBSA agrees that its written policy or policies should be clear and readily available to the public on EBSA's website.
(3) EBSA will include a reference to Regulations.gov as part of each NPRM webpage that includes public comments together with an explanation of its relation to Regulations.gov as a means to access public comments on EBSA's rulemaking initiatives. Internal and external users have indicated a preference to have public comments available on EBSA's website, among other reasons, because of the convenience of having the comments located in close proximity to the proposed rule and other related interpretive guidance. End users also have commented on and appreciated EBSA's logical and user friendly indexing and presentation of public comments, as compared to the docketing system contained in regulations.gov. EBSA, nonetheless, will raise GAO's recommendation informally with stakeholder groups as part of our evaluation of GAO's report and recommendation.
Recommendation 8
The Administrator of WHD should clearly communicate its policy for posting comments and their identity information, particularly for duplicate comments, to the public on the WHD website.

WHD response: WHD creates a Notice of Proposed Rulemaking (NPRM) webpage for all rules in which comments from the public are accepted. WHD will add the Agency Posting Guidelines to each NPRM webpage at the time of its creation to provide additional guidance to the public with respect to comment management.

### Appendix X: Agency Comments from the Securities and Exchange Commission



# Appendix XI: GAO Contact and Staff Acknowledgments

GAO Contact	Seto J. Bagdoyan, (202) 512-6722 or bagdoyans@gao.gov
Staff Acknowledgments	In addition to the contact named above, David Bruno (Assistant Director), Elizabeth Kowalewski (Analyst in Charge), Enyinnaya David Aja, Gretel Clarke, Lauren Kirkpatrick, James Murphy, Alexandria Palmer, Carl Ramirez, Shana Wallace, and April Yeaney made key contributions to this report. Other contributors include Tim Bober, Dahlia Darwiche, Colin Fallon, Justin Fisher, James Healy, Katie LeFevre, Barbara Lewis, and Maria McMullen.

GAO's Mission	The Government Accountability Office, the audit, evaluation, and investigative arm of Congress, exists to support Congress in meeting its constitutional responsibilities and to help improve the performance and accountability of the federal government for the American people. GAO examines the use of public funds; evaluates federal programs and policies; and provides analyses, recommendations, and other assistance to help Congress make informed oversight, policy, and funding decisions. GAO's commitment to good government is reflected in its core values of accountability, integrity, and reliability.
Obtaining Copies of GAO Reports and Testimony	The fastest and easiest way to obtain copies of GAO documents at no cost is through GAO's website (https://www.gao.gov). Each weekday afternoon, GAO posts on its website newly released reports, testimony, and correspondence. To have GAO e-mail you a list of newly posted products, go to https://www.gao.gov and select "E-mail Updates."
Order by Phone	The price of each GAO publication reflects GAO's actual cost of production and distribution and depends on the number of pages in the publication and whether the publication is printed in color or black and white. Pricing and ordering information is posted on GAO's website, https://www.gao.gov/ordering.htm.
	Place orders by calling (202) 512-6000, toll free (866) 801-7077, or TDD (202) 512-2537.
	Orders may be paid for using American Express, Discover Card, MasterCard, Visa, check, or money order. Call for additional information.
Connect with GAO	Connect with GAO on Facebook, Flickr, Twitter, and YouTube. Subscribe to our RSS Feeds or E-mail Updates. Listen to our Podcasts. Visit GAO on the web at https://www.gao.gov.
To Report Fraud,	Contact FraudNet:
Waste, and Abuse in	Website: https://www.gao.gov/fraudnet/fraudnet.htm
Federal Programs	Automated answering system: (800) 424-5454 or (202) 512-7700
Congressional Relations	Orice Williams Brown, Managing Director, WilliamsO@gao.gov, (202) 512-4400, U.S. Government Accountability Office, 441 G Street NW, Room 7125, Washington, DC 20548
Public Affairs	Chuck Young, Managing Director, youngc1@gao.gov, (202) 512-4800 U.S. Government Accountability Office, 441 G Street NW, Room 7149 Washington, DC 20548
Strategic Planning and External Liaison	James-Christian Blockwood, Managing Director, spel@gao.gov, (202) 512-4707 U.S. Government Accountability Office, 441 G Street NW, Room 7814, Washington, DC 20548